

Received July 18, 2019, accepted August 1, 2019, date of publication August 12, 2019, date of current version August 30, 2019. Digital Object Identifier 10.1109/ACCESS.2019.2933915

# **Relay-Assisted D2D Underlay Cellular Network Analysis Using Stochastic Geometry: Overview and Future Directions**

#### OLUWATOSIN AHMED AMODU<sup>[]</sup>, (Student Member, IEEE), MOHAMED OTHMAN<sup>[]</sup>,<sup>2</sup>, (Senior Member, IEEE), NOR KAMARIAH NOORDIN<sup>3</sup>, (Senior Member, IEEE), AND IDAWATY AHMAD<sup>1</sup>

<sup>1</sup>Department of Communication Technology and Network, Faculty of Computer Science and Information Technology, Universiti Putra Malaysia, Serdang 43400, Malaysia

<sup>2</sup>Laboratory of Computational Science and Mathematical Physics, Institute for Mathematical Research, Universiti Putra Malaysia, Serdang 43400, Malaysia <sup>3</sup>Department of Computer and Communication Systems Engineering, Faculty of Engineering, Universiti Putra Malaysia, Serdang 43400, Malaysia

Corresponding authors: Oluwatosin Ahmed Amodu (amodu\_oa@ieee.org) and Mohamed Othman (mothman@upm.edu.my)

This work was supported in part by the Malaysian Ministry of Higher Education through the Malaysian International Scholarship Scheme, and in part by the Malaysian Ministry of Education through the Research Management Center, Universiti Putra Malaysia, under UPM Journal Publication Fund, 9001103.

**ABSTRACT** Device-to-Device (D2D) communication is one of the enabling technologies for meeting the capacity requirements of the fifth-generation wireless systems (5G). It has diverse applications in traffic offloading, disaster management and content sharing, to mention a few. The network coverage and capacity further improve when relays are introduced to D2D communication. However, the interference becomes more severe since these devices also share resources with the traditional cellular users in the underlay. To take the benefits and avert the drawbacks of this spectrum sharing scenario, analytical tools capable of revealing the mathematical relationships among pertinent network design parameters are needed. This brings stochastic geometry (SG) into the picture. With SG-based analyses, designers can model concepts to understand, provide insights, and address the problems of spectrum sharing in relay-assisted D2D communication. Some of the key metrics of particular interest to network designers are the transmission capacity and spectral efficiency of D2D communication, as they reveal the performance gains and quantify the level of interference within the network. These enable them to properly correlate relevant cause-and-effect relationships before wealth and time are invested in network implementation. Despite the studies on the analysis of relay-assisted D2D underlay cellular networks using SG in recent years, there is no available survey material where researchers can find models, assumptions, key results and derived lessons to further comprehend this area and open up new research lines. This motivates the presentation of this paper which in addition to the aforementioned, gives elaborate discussions on promising areas for future research with respect to the recent advancements in D2D communication and SG research.

**INDEX TERMS** D2D, energy efficiency, relay, spectral efficiency, stochastic geometry, survey, transmission capacity.

#### I. INTRODUCTION

Device-to-Device (D2D) communication underlay cellular network infrastructure is receiving a lot of attention recently [1]. This paradigm facilitates mutual communication between devices in proximity with little or no intervention of the base station (BS) [2]. D2D communication has been identified as a promising technology for increasing the network capacity as well as user experience [3]. Additionally, it is a major prospective technology for attaining energy efficiency in cellular networks [4]. In future fifth-generation cellular networks (5G), D2D communication is regarded as an enabling technology to achieve high spectral efficiency (SE) and ultra-densification [5]. Furthermore, it enhances network throughput and reduces delay [6]. However, a major challenge in D2D networks within a cellular architecture is

The associate editor coordinating the review of this article and approving it for publication was Jie Tang.



FIGURE 1. Typical relaying scenarios for relay-assisted D2D communication.

co-channel interference that occurs when the existing cellular transmissions interfere with D2D communications.

The application use-cases of D2D communication become more restricted when the source and destination devices are not in proximity [7]. Hence, a universal solution cannot be achieved by D2D communication without the intervention of relays [8]. For example, when D2D pairs are farther apart, the outage probabilities of D2D links will increase. Furthermore, nodes might not be able to reach other nodes due to constraints in transmit power [9] in a single hop. In such cases, most traditional interference management schemes such as power control may not work. This motivates the need to enhance the range of communication, thereby reducing the outage probability [10].

In relay-assisted D2D communication, the source and destination devices discover an idle node between them to establish mutual communication. The source node transmits data to the relay which decodes and forwards it to the destination node. Some typical relaying scenarios are illustrated in Fig. 1. Relay-assisted D2D communication has several use cases. According to [13], the ubiquity of D2D communication makes the incorporation of multihop D2D communication conceivable as a part of a future standard. This is because multihop D2D communication is not bounded to specific geographic locations like the traditional D2D communication.

Device relaying has several other benefits. It can inherently bring to light the potential of cooperative communication to meet the growing demand for higher data rates and capacity in 5G cellular systems [12]. The use of relays facilitates multihop transmission which improves spectral reuse within the network. Asides the potential of achieving higher data rates and improving the network capacity, there are also 'promises' of improved quality of service (QoS), and network load balancing. Also, relays can be used to improve network performance in poor coverage areas [11], such as places with damaged cellular infrastructure [13].

In view of the above significance of relays, this paper reviews the performance analysis of relay-assisted D2D communication in cellular networks. We concentrate on stochastic geometry (SG)-based analysis of the network as it has recently been proven to be effective for understanding interference in spectrum sharing networks. Furthermore, we discuss promising future directions that have emerged due to recent development in SG research, such as signal-tointerference ratio (SIR) meta-distribution, spatio-temporal interference correlation, and the use of generalized fading channels. Moreover, this paper also presents a number of research opportunities peculiar to relay-assisted D2D communication. Note that some of the future directions presented in this paper are also applicable to other wireless communication systems.

#### A. THE ROLE OF STOCHASTIC GEOMETRY IN NETWORK ANALYSIS

Stochastic geometry is a sophisticated tool from mathematics which gives insights into spatial averages taken over several

TABLE 1. Abbreviations and mean	ings.
---------------------------------	-------

Abbreviation	Meaning
3GPP	Third-generation partnership project
5G	Fifth-generation cellular network
ASR	Average sum rate
BS	Base station
CRN	Cognitive radio network
CUE	Cellular user equipment
D2D	Device-to-Device
DUE	D2D user equipment
FDAF	Full duplex amplify and forward
FDMA	Frequency division multiple access
IC	Interference cancellation
LOS	Line-of-sight
LTE	Long-term evolution
LTE-A	Long-term evolution advanced
PCP	Poisson cluster process
PDF	Probability density function
PPP	Poisson point process
QoS	Quality of service
RN	Relay node
SE	Spectral efficiency
SG	Stochastic geometry
SINR	Signal-to interference-plus-noise ratio
SIR	Signal-to-interference ratio
STP	Successful transmission probability
TDD	Time division duplexing
TC	Transmission capacity
UAV	Unmanned aerial vehicle
UE	User equipment

realizations for a large number of nodes at various locations. These insights could be in the form of outage probability, interference, data rate and signal-to-interference-plus-noise ratio (SINR) [14], [16]. SG can be employed to design spatial stochastic models for devices' locations.<sup>1</sup> Additionally, point processes can be generalized in two or three dimensions and indexed by time [15]. SG is meritorious as it can be used to characterize the SINR in cellular, sensor and adhoc networks. Traditional approaches in communication theory can not be easily deployed for that purpose as interference in such networks highly depends on the network geometry. Another reason is that these systems involve a high level of uncertainty. For instance, it is very hard for each node to know the location and predict the channel state of all other nodes [14]. Thus, statistical models depicting the relationship of devices in space using SG can be used to answer questions such as [15]:

- In the next decade, what effects will the large-scale deployment of unplanned infrastructure have on the spectrum and its users?
- What approach could be used to design wireless multihop networks which may involve mobility and randomly scattered users?

• How can the networks' throughput/capacity and connectivity/coverage be determined?

#### **B. RELATED SURVEYS**

This subsection briefly describes the contributions of related surveys/tutorials on SG for modelling various aspects of wireless communication. It is worth noting that D2D (and relay-assisted D2D) communication research using SG started becoming prominent after the first few surveys emerged.

Towards understanding the role of SG in modelling interference, Haenggi *et al.* [14] presented a seminal survey article on the use of SG and random graphs for the analysis of wireless networks in 2009. They expounded on SG based techniques such as point process, boolean model, percolation and connectivity. Insights were given into the history of SG, interference characterization and applications such as routing, secrecy, information propagation as well as point processes with fading.

In 2010, Andrews *et al.* [15] presented an entry-level tutorial illustrating the use of spatial models and relevant analytical techniques for wireless networks design. They derived metrics such as coverage and connectivity, throughput and end-to-end rate in wireless networks. The role of SINR and its constituent parameters and 'building blocks' were discussed. Furthermore, applications of spatial models to adhoc networks, femtocells and cognitive radio were also examined.

By 2013, Elsawy *et al.* [16] comprehensively surveyed and classified SG models in literature for single, multi-tier and cognitive cellular networks. The authors presented a novel taxonomy of the techniques used by researchers in SG to analyze the network performance with examples and references. Furthermore, baseline SG models and the use of SG for modelling technologies such as MIMO were also opened up. No discussion on the modelling of D2D communication using SG appeared in the paper.

In 2016, Andrews *et al.* [17] presented a primer on modelling and analyzing cellular networks using SG. They introduced point processes and tools for calculating the coverage probability metric showing their applications in both downlink (single/multi-tier) and uplink scenarios. Later in the year, Elsawy *et al.* [18] presented a tutorial on this subject including an exact baseband interference characterization and analysis of error rate in cellular networks. A unified approach for analyzing several pertinent metrics such as outage probability, error probability, and the ergodic rate was given. The paper described the significance of SG to D2D research briefly.

From the aforementioned, we observe a clear gap in the existing literature on the detailed exposition on SG models, methods and metrics for relay-assisted D2D communication studied in the last few years. A comparison of these previous surveys in Table 2 shows that a survey discussing SG-based analysis for D2D is absent in the literature, let alone relay-assisted D2D communication which is the focus of this paper.

<sup>&</sup>lt;sup>1</sup>Network designers can perform interference characterization using such models and provide insights into the network performance in different physical conditions based on diverse metrics.

#### TABLE 2. Related surveys on SG and their focus.

	Acapacto	References						
	Aspects	Haenggi et al. [14]	Andrews et al. [15]	Elsawy et al. [16]	Andrews et al. [17]	Elsawy et al. [18]		
	Background on PPP	1	1		1	1		
s	Interference characterization/Outage		1		1			
Analysi	Techniques for Analysis			$\checkmark$				
	Downlink Analysis				✓			
	Uplink Analysis				<ul> <li>✓</li> </ul>			
	Scaling laws	<ul> <li>✓</li> </ul>						
	Connectivity							
	Coverage		√		$\checkmark$			
cs	Capacity	√						
etri	Outage probability	$\checkmark$			$\checkmark$	$\checkmark$		
e M	Transmission Capacity	$\checkmark$	$\checkmark$					
anc	Ergodic capacity					$\checkmark$		
	Connectivity	$\checkmark$						
erfc	Percolation	$\checkmark$						
P	Symbol error probability					$\checkmark$		
	Pairwise error probability					$\checkmark$		
	Throughput		$\checkmark$					
	Single cell networks			$\checkmark$	$\checkmark$	$\checkmark$		
ons	Heterogeneous Networks			$\checkmark$	$\checkmark$	$\checkmark$		
cati	Cognitive Networks			$\checkmark$	$\checkmark$			
plid	MIMO			$\checkmark$	$\checkmark$	$\checkmark$		
Υİ	mmWave				$\checkmark$			
	General	$\checkmark$						
~	Biasing and load balancing			$\checkmark$	$\checkmark$	$\checkmark$		
ect	Spectrum Allocation			$\checkmark$				
asp	Frequency-reuse			$\checkmark$		$\checkmark$		
ing	PPP approximation					$\checkmark$		
label	Error Analysis					$\checkmark$		
Mc	Ergodic rate analysis					<ul> <li>✓</li> </ul>		
	Gaussian Signalling Approx.					<ul> <li>✓</li> </ul>		

#### C. MOTIVATION

Within the last decade, the stringent QoS requirements posed by future generation networks coupled with the data-hungry applications continue to push experts towards developing innovative technologies to accommodate 5G network requirements. Paradigms such as heterogeneous networks, ultra-dense networks, cloud radio access networks (C-RANs) and edge computing have emerged. Amongst such paradigms is D2D which facilitates the communication of devices without traversing the cellular BS. This differentiates it from traditional cellular communication. Not only has much been envisioned concerning technologies, but modelling techniques have also received a share of researchers ingenuity and novelty. New techniques in optimization, game theory and particularly SG arrive prestigious publication venues time after time.

Looking from the perspectives of technologies and techniques described above, we realize that relay-assisted D2D has positioned itself as a potential approach to tackle some of the traditional challenges of D2D communication such as coverage. Also, efforts have been made to model and unveil its performance in diverse application use cases and physical layer configurations using mathematical tools such as SG. However, as much as spectrum sharing systems have been researched using SG, there is no survey on relay-assisted D2D communication till now (see Table 2 and Fig. 2). In this regard, we study the models and assumptions on this subject and derive research-based lessons from this area. We thought this is not sufficient since much is yet to be tapped from the research of other spectrum sharing systems. So we critically derive lessons from other wireless systems in terms of models and techniques.

Furthermore, we tried to answer new questions that arise as the standardization of this (relay-assisted D2D) technology might be very close. For example, what are the common techniques and approaches currently used? What are the recent trends that would likely receive a large share of researchers time in years to come? Besides, we look into some of the integrand technologies which D2D could also 'sail over' to 'give birth' to the future 5G network where user demands are tactfully satisfied using the best of technologies now available and envisioned. In this respect, we explore several new SG-based domains and aspects relating to relay-assisted D2D communication. This paper also reveals some of the current models, findings, and future directions in this regard.

<b>—</b>	-General issues and emerging aspects
T	$\downarrow$ Underlay and overlay D2D [19] $\heartsuit$
	Cellular-based D2D research and prototypes [20] ¤
	Research issues and under-explored areas [21] ¤
	D2D and its merger technologies [22] ¤
	Control and data planes separation architecture [23] ¤
	D2D w.r.t 3GPP standardization [24] ¤
	D2D mobility w.r.t practice and challenges [25] ¤
	$\square$ D2D and other offloading techniques [26] $\square$
	Features and use cases [27] ¤
	Socially-aware D2D communication [28] ¤
	Socially-driven D2D and research gaps [29] ¤
	D2D channel models [30] ¤
	$\square$
	$\square$
	mmWave propagation for D2D [33]
	NOMA-based D2D for 5G and IoT [34]
	D2D in ultra-dense networks and research challenges [35] ¤
	D2D device discovery and research issues [36] ¤
	D2D aspects, recent advances, and challenges [37] ¤
P	-Security
	Architecture and Security [6] ¤
	Security and privacy in D2D communication [38] $\approx$
	Security architecture in SOPP LTE/LTE-A [59]
	3GPP LTE D2D security architecture and requirements [40]
L	Free Review and classification of security solutions [41]
	Methode in the service and security [42] ¤
	Resource allocation and mode selection [43] ¤
	Resource allocation optimization [44] $\alpha$
	Distributed resource allocation schemes [45]
	Resource allocation schemes and challenges [46]
	Resource allocation, power conservation and security [47] ¤
	Interference alignment [48] $\heartsuit$
	Frequency and level of control [49] ¤
L	– Multihop
	Routing [13] ¤
	Intelligent D2D and routing [50] ¤
	Cooperative communication [51][52]
	Relay selection techniques [53]
	Relay-assisted D2D analysis using SG ¤ This paper

**FIGURE 2.** Taxonomy of D2D surveys. • indicates journal publication.

#### **D. CONTRIBUTIONS**

In this paper, we present the following contributions:

- Classification and discussion of existing literature on relay-assisted D2D communication network performance using SG in the context of metrics such as transmission capacity (TC) and SE.
- Derivation of lessons from reviewed literature and other spectrum sharing systems to guide readers to potential areas for future considerations.
- Exposition on open areas through the recent breakthrough in SG research as well as developments in D2D, relay-assisted D2D, and wireless communications

in general. Emerging research areas like SIR metadistribution, where SG is highly propitious are identified.

The rest of this paper is organized as follows: Section II introduces D2D communication modes and spectrum sharing for relay-assisted D2D networks. It also explains some relevant performance metrics used in SG-based analysis of relay-assisted D2D networks. One of the most commonly studied metric, transmission capacity, is further discussed in Section III. Section IV reviews other performance metrics in SG-based analyses of relay-assisted D2D such as spectral efficiency, transmission rate, and service probability. Section V presents the potential areas for future work and further directions on relay-assisted D2D communication. Section VI concludes this paper. Table 1 provides the meanings of abbreviations used in this paper.

#### II. RELAY-ASSISTED D2D SPECTRUM SHARING MODELS AND RELEVANT PERFORMANCE METRICS

This section first introduces the D2D communication modes. Afterward, spectrum sharing models in relay-assisted D2D networks and some performance metrics studied in the network analysis are discussed.

#### A. D2D COMMUNICATION MODES

Considering how user devices can access the licensed or unlicensed spectrum, D2D communication can be classified into in-band (licensed) and out-band (unlicensed).<sup>2</sup> The in-band D2D can be further categorized into underlay and overlay modes (see Fig. 3). In the underlay mode, devices use the same spectrum for cellular and D2D communication i.e., they share the same radio resources. One motivation for the consideration of inband D2D communication is that the licensed spectrum has a considerable level of control compared with the unlicensed band [6]. Asides this, [6], [19], [55] highlighted a number of benefits which include improved spectral efficiency due to the exploitation of spatial diversity. Also, cellular devices can support inband D2D since the cellular frequency will be utilized. Furthermore, resource allocation techniques can be easily deployed to manage the network's OoS.

Despite these benefits, inband D2D communication has its own drawbacks [43], [55]. A typical challenge in underlay D2D is interference which could be addressed when proper power control and interference management techniques are considered. However, these may require very complex resource allocation techniques [19]. In the overlay mode, D2D uses a dedicated spectrum orthogonal to that of cellular communication. However, the cellular resources could be underutilized as a typical user cannot perform simultaneous cellular and D2D transmission. The use of the overlay mode may poorly affect the QoS due to a high level of unman-



FIGURE 3. In-band D2D spectrum sharing modes.

aged interference resulting from other wireless technologies sharing the same spectrum [1], [55]. This interference cannot be controlled by the BS. Thus the underlay D2D is more popular [19] and as such, it is the focus of this paper.

Considering relay-assisted D2D communication, user devices can function in one of three modes: a cellular mode where they function as traditional CUEs, D2D mode where they communicate in a D2D fashion, and relay mode where they relay data for other devices.

#### B. SPECTRUM SHARING IN RELAY-ASSISTED D2D NETWORKS

In the quest for effective utilization of the scarce cellular network resources, it is imperative that the wireless channel is shared efficiently [56]. Spectrum sharing occurs when users/radio communication systems use the same spectrum resource. Sharing could occur in frequency, time and place (space) [57]. A fundamental component which affects spectrum sharing systems is the device density. In a dense relay-assisted D2D communication network, the chance of a more efficient spatial re-use is higher since the spectrum will be 'heavily' used by these systems across the entire network. Note that for effective network performance, the QoS requirements of participating systems have to be met [58]. In this regard, the level of reliability in terms of success probability for specific data rates has to be given proper consideration.

On the other hand, the demand for spectral resources in dense relay-assisted D2D networks becomes higher. Also, such scenarios experience more interference which significantly affects network performance. The level of interference dictates what type of resource allocation mechanisms, mode control, and power control is required to guarantee the target SINR thresholds for different systems within the network. Thus, a deeper understanding of the effect of device density, as well as other design parameters on the network, is crucial.

There are several models for spectrum sharing in relay-assisted D2D networks consisting of cellular devices, D2D devices, and relay nodes. A pictorial view of a typical network model where idle D2D devices can function as relay nodes is shown in Fig. 4. The next subsection describes key performance metrics used in this context.

 $<sup>^{2}</sup>$ In out-band D2D communication, the interference from D2D to eNodeB-UE is absent [54] as the frequency spectrum used by D2D does not overlap with the cellular spectrum [19].



**FIGURE 4.** A pictorial view of a typical network model involving cellular, D2D, and relay nodes.

#### TABLE 3. Notations and meanings.

Notation	Meaning
$\lambda_c$	Density of CUEs
$\lambda_d$	Density of DUEs
$\lambda_u$	Density of UEs
$\lambda_r$	Density of RNs
$\lambda_{BS}$	Density of BSs
$\Phi_c$	Homogeneous PPP of CUEs
$\Phi_d$	Homogeneous PPP of DUEs
$\Phi_u$	Homogeneous PPP of UEs
$\Phi_r$	Homogeneous PPP of RNs
$P_c$	Cellular transmit power
$P_d$	DUE transmit power
$P_r$	RN transmit power
$P_{re}$	RN existence probability
$T_c$	CUE SIR threshold
$T_d$	DUE SIR threshold
$T_r$	RN SIR threshold
$\alpha$	Path loss coefficient
R	D2D Link distance
$d_{n0}$	Expectation of D2D Link distance
ρ	Power allocation coefficient

#### C. PERFORMANCE METRICS

Some performance metrics used in the literature to evaluate relay-assisted D2D networks are described below. For simplicity, we have excluded mathematical definitions for these metrics. Interested readers could refer to the references cited for these definitions. The notations used in this paper are defined in Table 3.

#### 1) TRANSMISSION CAPACITY

This refers to the number of successful transmissions which take place per unit area in a network, given a constraint in the outage probability [59]. TC measures the intensity of transmissions in space. It is motivated by fixing the outage probability to achieve a proper characterization of network performance. TC metric was initially developed for the analysis of spread spectrum in adhoc networks. It has found a lot of applications in wireless networks (specifically in decentralized networks which are difficult to characterize) since then. TC has been studied with respect to systems such as interference cancellation (IC) [60]–[62]; scheduling and power control [63], [64]; cognitive radio network (CRN) [65]–[67]; frequency spread spectrum [68]–[70]; as well as multiple antennas [59], [71]–[73]. Section III reviews literature on TC analysis of D2D and relay-assisted D2D communication using SG.

#### 2) SPECTRUM-SHARING TRANSMISSION CAPACITY

As compared with TC which is an indication of the maximum number of successful transmissions [62], [68], spectrum-sharing TC applies to several systems with an arbitrary spatial density set within a range, where all outage probability constraints are to be satisfied. In this case, it refers to the number of successful transmissions [62].

#### 3) ACHIEVABLE TRANSMISSION CAPACITY

For spectrum sharing systems, the achievable TC can be used to verify the impact of the secondary systems spatial density on the network capacity. Contrary to the TC, the achievable TC should satisfy the outage probabilities of all systems.<sup>3</sup> It thus includes the power ratio of coexisting systems as well as their multiple outage probability constraints [74]. Moreover, this is considered with respect to Shannon's formula. In other words, the spectrum sharing transmission capacity is multiplied by  $log(1 + T_n)$  where  $T_n$  is the target SINR threshold of system *n* (see [10] for example).

#### 4) AREA SPECTRAL EFFICIENCY

Area spectral efficiency quantifies the SE of cellular systems having a variable rate transmission. It can be referred to as the summation of the highest average data rates for a unit bandwidth, in a unit area supported by a cell's BS [75]. This definition encapsulates the trade-off between the SE for the cellular system, user link, and the quality of the communication link provided to users. Subsection IV-A discusses some of the works that studied SE in relay-assisted D2D communication using SG.

Next, we examine relevant literature considering SG-based analysis for evaluating relay-assisted D2D underlay cellular system performance. We present a classification of the reviewed literature based on their performance metric and assumptions as shown in Fig. 5.

#### **III. TRANSMISSION CAPACITY ANALYSIS**

As stated in the previous section, transmission capacity is a fundamental performance metric that has frequently appeared in SG-based D2D communication research. In this section, we review relevant works that used SG for TC analysis of D2D communication with and without relays. Lessons learned from the review are discussed and some open issues are highlighted.

<sup>&</sup>lt;sup>3</sup>In other words, primary and secondary systems in the context of CRNs.



FIGURE 5. Taxonomy of reviewed literature on relay-assisted D2D underlay cellular networks based on performance metrics (layer 1), assumptions and models (layer 2) used.

### A. TC ANALYSIS OF D2D COMMUNICATION WITHOUT RELAYS

D2D communication can provide higher capacity in terms of sum rate and increased throughput when compared with the traditional cellular communication [95], [96]. In this subsection, we briefly discuss methods and findings of relevant SG research that have studied TC metric of D2D communication without relays. Table 4 highlights some of their key aspects.

A premier investigation on the extent to which capacity improvement can be achieved in cellular systems when D2D modes are used is presented in [94]. The authors determined how device user density impacts the network's capacity. In this regard, the optimal density of D2D users that maximize the capacity of D2D-enhanced networks was derived. To provide insights necessary for adequate resource allocation and QoS enhancement in D2D networks, Kachouh *et al.* [76] investigated different aspects of LTE-based D2D communication using a convex-concave computational optimization technique to derive optimal ratio of transmit power. They recorded that an optimal power allocation solution exists for D2D which maximizes the global capacity of the Long-term evolution (LTE) network.

Lin *et al.* [79] studied how a typical device should choose its communication with a co-device or BS. They proposed a hybrid and tractable network model where mobile nodes were modelled using the Poisson point process (PPP) and the base stations were placed based on a hexagonal grid model. They assumed channel inversion power control where the transmit power of a cellular or D2D user,  $P_i = L_i^{\alpha}$ .  $L_i$  denotes the length of radio links. They also assumed that the average received power is 1 due to channel inversion. This implies that a proportionality coefficient  $\phi$  (where  $\phi \ll 1$ ) exists which maps the transmit power of user equipments,  $P_i$  to an actual

TABLE 4.	Key aspects of th	e reviewed	literature or	1 capacity	analysis of
D2D com	nunication witho	ut relays.			

	Aspects	References							
	Aspecis	[76]	[94]	[77]	[78]	[79]			
	Transmit power ratio	$\checkmark$							
ion	Power allocation ratio				$\checkmark$				
zati	Capacity optimization	$\checkmark$	$\checkmark$	$\checkmark$					
ni	Transmission density			$\checkmark$	$\checkmark$				
ptii	D2D mode			$\checkmark$	$\checkmark$				
Ō	D2D density					$\checkmark$			
	OP threshold			$\checkmark$	$\checkmark$	$\checkmark$			
	ST Probability		$\checkmark$	$\checkmark$	$\checkmark$				
ng	Poisson Point Process		$\checkmark$	$\checkmark$	$\checkmark$				
elli	Channel Inversion					$\checkmark$			
po	Spectrum access factor					$\checkmark$			
Σ	Single-cell scenario	$\checkmark$	$\checkmark$						
	Multi-cell scenario		$\checkmark$						

transmit power  $\tilde{P}_i$ . Their findings show that a trade-off exists between the spectrum access factor and D2D mode selection threshold for choosing either direct link communication or traditional cellular communication.

#### B. TC ANALYSIS OF RELAY-ASSISTED D2D COMMUNICATION

In this subsection, we review relevant works that used SG for TC analysis of D2D communication with a focus on relay-assisted D2D underlay cellular network. We describe a typical spectrum sharing model for the reviewed articles and highlight their key aspects in Table 5.

#### C. REFERENCE MODEL

This model (see Fig. 6) uses the generalized successful transmission probability (STP) for all links within the network and assumes that devices have the same transmission power for equal transmit distances. Also, D2D users are distributed based on PPP. The representation of devices is similar to that in Fig. 4. The generalized form of this model was presented by Lee *et al.* [58] as an extension to the TC metric proposed by Weber *et al.* [59] for spectrum sharing systems. It has been widely used in research such as [10], [80]–[82], [88], [97]. For '*n*' spectrum sharing systems, the SIR at a typical receiver in system *n* is given as

$$\operatorname{SIR}_{n} = \frac{P_{n}\delta_{n0}d_{n0}^{-\eta}}{\sum_{\forall i \in \{n\}}I_{i}},\tag{1}$$

where  $I_j = (\frac{P_j}{P_n}) \sum_{k \in \Psi_j} \delta_{jk} |X_{jk}|^{-\eta}$ .  $\delta_{jk}$  and  $X_{jk}$  refer to the Rayleigh fading coefficient and

 $\delta_{jk}$  and  $X_{jk}$  refer to the Rayleigh fading coefficient and the distance to interfering node from the origin k,  $(k \in \Psi_j)$ in  $S_j$ ,  $j \in \{c, d, r\}$ , respectively.<sup>4</sup> Similarly,  $\delta_{n0}$  and  $d_{n0}$  are the Rayleigh fading coefficient of the reference transmitter to the typical receiver and their mutual distance, respectively. In other to have reliable data transmission, the systems SIR must satisfy the minimum SIR requirement i.e., SIR<sub>n</sub>  $\geq \theta_n$ , where  $\theta_n$  is the SIR threshold of system  $S_n$ .

 $^{4}c$ , d, r denote cellular, D2D, and relay user equipment, respectively.

	Aspects	References									
	Aspects	[80]	[81]	[10]	[82]	[88]	[83]	[87]	[85]	[84]	[86]
	Optimal relay users distribution			<ul> <li>✓</li> </ul>							
ion	D2D system optimization				<ul> <li>✓</li> </ul>						
otimiza	Optimal capacity								$\checkmark$	$\checkmark$	
	Optimal transmit power				✓						
o	Optimal relay selection strategy								$\checkmark$		
	Optimal transmit power ratio			$\checkmark$	$\checkmark$						
	Poisson point process	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
	Random relay Selection	✓	$\checkmark$			$\checkmark$					
JS	Nearest relay Selection										$\checkmark$
lioi	Random relay distribution			$\checkmark$	$\checkmark$						
du	Full duplex relays						$\checkmark$			$\checkmark$	
Asst	Rayleigh fading	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$						$\checkmark$
	Nakagami-m fading							$\checkmark$			
	Direct link communication option	<ul> <li>✓</li> </ul>	$\checkmark$	<ul> <li>✓</li> </ul>	$\checkmark$						$\checkmark$
	Constant data rate assumption			✓							
	Distance expectation	$\checkmark$	$\checkmark$								
	Achievable TC			$\checkmark$	$\checkmark$						
ect	Two relay hops	<ul> <li>✓</li> </ul>									$\checkmark$
Asp	Single-cell scenario								$\checkmark$		
ne 7	Multiple relay hops			$\checkmark$	$\checkmark$						
niq	Caching strategy										$\checkmark$
	Social-aware							$\checkmark$			
	Spectrum Management				$\checkmark$						
	Successful transmission probabilities	$\checkmark$	$\checkmark$	$\checkmark$							$\checkmark$
	Closed form expression	<b>√</b>	$\checkmark$	$\checkmark$	<ul> <li>✓</li> </ul>						

TABLE 5. Key aspects of the reviewed literature on capacity analysis of relay-assisted D2D communication.



**FIGURE 6.** A reference model that shows relay nodes in the intersection region between two D2D users.

The STP used in this model is given in Lemma 1 in the appendix where it is narrowed down to three (n = 3) systems (i.e., cellular devices, D2D devices and relays) suitable for the relay-assisted D2D framework as deployed in [80], [81], [83].

#### D. RELATED LITERATURE

#### 1) HALF-DUPLEX RELAYS

Using the above model, [80] studied the TC of relay-assisted D2D communication in underlay mode with respect to transmission link distance. This was extended in [81] to both

underlay and overlay modes. The authors used two relaying approaches (i.e., relay only, and either relay or direct link) to enhance the performance of D2D communication. They presented closed-form expressions for the TC of D2D communication, derived capacity gains, and also considered variable D2D link distance (which is more practical) based on these modes. Wen *et al.* [82] studied a multihop D2D network by analyzing the achievable TC over Rayleigh fading channel. The probability density function (PDF) of SINR in the spectrum sharing network was formulated. An important assumption made is that the distance traversed by relay transmissions decreases with the increase in the number of hops. Their study reveals that the power ratio of the cellular and D2D system can influence the interference level and system capacity.

Lin *et al.* [10] investigate relay-assisted D2D communication to solve the problem of an increase in outage probability which occurs when D2D pairs are far apart. In the model, an idle UE is used to establish a link for communication between the source UE and destination UE as shown in Fig. 7 where  $\frac{R^2}{2} + 2r_h^2 = r_1^2 + r_2^2$ . With an assumption of decode and forward relay protocol, the outage probability is given as

$$P_r^o = 1 - P\{\gamma_1 > \gamma_d\} P\{\gamma_2 > \gamma_d\},\tag{2}$$

where  $P_r^o$  is the D2D outage probability,  $\gamma_1$  is the target SINR of the first hop and  $\gamma_2$  is the target SINR of the second hop. Similarly,  $\gamma_d$  is the target D2D SINR. Having assumed a



FIGURE 7. The topology of relay-assisted D2D communication.

random distribution of RNs, the authors analyze the optimal relay users distribution, PDF of the optimal relay users' position and optimal ratio of transmit power in terms of cellular and D2D users' densities. These results were used to present a theoretical analysis of the achievable TC of the network. They conclude that when the "wireless environment" is poor (particularly when users are densely distributed and the mutual distances between devices in D2D pairs extend), the achievable TC of D2D network can be enhanced by relay-assisted transmission.

#### 2) FULL-DUPLEX RELAYS

Although half-duplex relaying is considered a very practical system [98], it is inefficient with regards to resource utilization since relays cannot transmit and receive on the same channel [99]. The authors in [83] considered full duplex amplify and forward (FDAF) relay nodes to assist cellular and D2D communication. These relays can transmit and retransmit signals thereby improving the network coverage and SE, amplifying the signal power, and achieving lower delay within the system. Devices can either communicate directly or through the FDAF relays. CUE, DUE and RNs were modelled as a PPP and D2D users share the uplink with cellular users. The paper analyzed the coverage probability and TC for both cellular and D2D links in this context. The results indicate that both coverage and capacity of cellular and D2D links are affected by parameters such as the RN density, mutual distance of D2D pair and D2D density. Although the introduction of FDAF within this framework is quite unique, the scenario can be adapted to social-aware relaying where devices require incentives that motivate them to relay. Furthermore, the use of full duplex relays can be studied under generalized fading channels since Rayleigh fading was assumed in the paper.

Delgado and Labeau [84] investigated the issues of fairness and network capacity in mobile relay-enhanced cellular networks. Under the presumption that the channel capacity at the center is usually greater than that of the cell edge, they studied the use of full duplex relays by investigating how fairness and overall network capacity can be maximized. A greedy algorithm selects the RNs which increases the channel capacity at the cell edge and reduces the transmission power consumption by 80%. The algorithm also improves the Jain fairness and average network capacity by 8% and 20%, respectively. However, the investigation in this paper is restricted to a single cell scenario.

#### 3) MULTI-CELL NETWORK SCENARIO

Feng et al. [7] consider a tractable model for heterogeneous multi-cell network where idle UEs can function as RNs with interference and spatial distribution taken into account. To achieve this, they derive a closed-form expression for the coverage probability using three different location-aware relay selection schemes: all relay selection, where all the idle UEs can function as potential relays; sectorized relay selection, where the selected relay-to-destination distance is shorter than the source-to-destination distance; and distance-based relay selection, where the source can obtain its distance to relay nodes using GPS or the Received Signal Strength Indication. Therefore, only idle nodes within a specified radius can be selected. Results show that relay-assisted transmission can improve D2D network performance in terms of both TC and coverage probability. Also, distance-based relay selection provides the highest D2D TC and maintains the coverage probability compared to the all relay selection mechanism. Thus, it was concluded that the distance-based scheme was the best for cooperative D2D communication which underlays cellular networks.

#### 4) MULTIPLE SPECTRUM SHARING

Lin *et al.* [85] investigated the achievable TC for D2D communication in both underlay and overlay direct and relay-assisted D2D modes. They analyzed an optimal selection approach between single and multiple sharing methods. Using the constraint of minimum energy consumption, the optimal relay selection strategy in terms of relay-assisted D2D communication was obtained. Results showed that the proposed optimal relay selection strategy lowers the energy consumption of relay-assisted D2D communication in underlay mode.

#### 5) CACHING

The aim of caching popular files in a D2D group is to reduce the burden of the BS. This is achievable since the files requested can be catered for by the D2D links. Although the BS burden is relieved, it has to cache all the files as the D2D caching group refreshes at intervals of time. As such D2D communication would also share channel resources with cellular links and the interference that results from several D2D links would impact the quality of cellular links. To address this, Fan *et al.* [86] fix a threshold SINR at the BS so that the quality of cellular links can be guaranteed. They provide a mathematical model for the network capacity of cellular, D2D and relay links, and propose a distributed caching strategy where the most popular files within a D2D group are cached to improve the performance of the network in terms of capacity. This is based on the rationale that the popularity



FIGURE 8. The system model and spectrum access scheme for [87].

of a file indicates its level of demand by mobile devices. The popularity of given k files is expressed as

$$q_{i} = \frac{\frac{1}{i^{\gamma}}}{\sum_{j=1}^{k} \frac{1}{j^{\gamma}}}$$
(3)

where  $q_i$  is the probability that the *i*th file is requested by a typical user.  $1 \le i \le k$  and  $1 \le j \le k$ ,  $q_1 \le q_2 \le \cdots \le q_k$ . It is assumed that a relay link is used when the distance between the requesting user and the BS is greater than a distance threshold. Numerical simulations were used to show that the proposed architecture significantly improves the network capacity.

#### 6) SOCIAL-AWARE

Establishing relay links can be very challenging as RNs may not be willing to relay data for other users due to their selfish nature [87]. A typical relay user might wonder: "what benefit does one get by relaying data for an unknown user?" This is an aspect with little attention in SG-literature. Considering the above, Chun et al. [87] incorporated this social aspect of D2D nodes into the relay-assisted D2D framework and modelled the decision on being a relay node using the 'social comparison' presented in [100]. They allocated a specific portion  $\beta$  of the spectrum to D2D communication, while the complementary portion  $1 - \beta$  is allocated to the cellular spectrum (where  $0 \le \beta \le 1$  as shown in Fig. 8). Specifically, the authors formulated and characterized a donation game with the likeliness that devices would cooperate within an evolutionary framework. They studied a practical case where the cooperation probability is less than 100% cooperation (i.e the ideal case) and observed a lower TC and higher outage probability in the non-ideal case. Although after an adequate number of evolutionary generations, the cooperation probability evolved and the transmission performance approaches that of the full cooperation. This reveals that scenarios exist where device cooperation can evolve with time without any central enforcement.

#### 7) CU-ASSISTED TRANSMISSION

In [88], the authors analyzed D2D TC when the cellular users allocate a certain portion of their power to assist in mutual



**FIGURE 9.** Model of D2D communication assisted by cellular users [88] © IEEE.

D2D communication while keeping the quality of their own cellular transmissions intact. Under this assumption, if a number of available cellular users exist within the communication range of D2D devices, a cellular device is selected randomly in every time slot to assist the D2D transmission using time division duplex (TDD). The transmission power of the assisting cellular user is modelled as  $\rho P_c$ , where  $P_c$  is the transmit power of the cellular user and  $0 \le \rho \le 1$ . Also, the expectation of the distance between the cellular user and D2D receiver (see Fig. 9) is given as  $\frac{2}{3}R$ , where *R* is the maximum communication range of D2D link. The average number of D2D users requiring the assistance of one cellular user is  $\lambda_1 \pi R^2$ , where  $\lambda_1$  is the density of D2D users. Thus, the assisting cellular user uses  $(1 - \lambda_1 \pi R^2 \tau)P_c$  to 'keep' its cellular transmission.

#### 8) ERGODIC CAPACITY

In D2D underlay networks, [89] studied the trade-offs between ergodic capacity and power consumption to discover whether the merits of D2D underlay surpass its demerits when integral system aspects are considered under Rician fading assumption. These aspects include cellular BS density, DUE transmit power and device density. With the assumption that the D2D pairs are randomly distributed, the authors derived the ergodic capacity of the device and cellular users in closed form and further investigated the device density and transmit power required to achieve minimal energy consumption. The ergodic capacity was maximized using a two-stage scheme. This led to the presentation of a D2D framework for demonstrating the expedient system parameters to obtain the highest capacity gains. An interesting aspect of this work is that the ergodic capacity was evaluated considering the D2D distance as a random quantity i.e., not fixed, which is more practical. Note that, when the location of D2D transmit users were assumed to be dynamic, such users were deemed to move with the same velocity.

#### E. LESSONS LEARNED

Here, we derive lessons from the literature examined in this section. Insights are also drawn from some other networks,

such as CRN which is an emerging merger technology with D2D communication.

From the literature, it was learned that both SG and optimization tools can be combined to effectively model, analyze and derive optimal values for relevant parameters and metrics in a relay-assisted D2D network. For instance, parameters such as global capacity, selection constraint for D2D pairs and spectrum sharing mechanisms can all be optimized. We note that the proposed optimization should depend on the nature and number of parameters to be optimized within the network. The assumption of a random selection of relays to forward data between D2D pairs in [80], [81] may not capture the variety of use-cases of D2D communication which might require different packet forwarding schemes (e.g., mission-critical applications). Hence, other relay node distribution should be considered with application target in mind. Similar to point processes having varying suitability based on medium access protocol or target application, appropriate and optimal relay selection strategies are also required.

The PPP is a common assumption which has been widely used in the study of D2D systems. It is important to cite that other point processes are more practical in scenarios where devices are not randomly scattered within the studied regions. Examples of such are: the Poisson Cluster Process (PCP) [101], [102], combination of PCP and Poisson Hole Process [103], binomial point process [104] and the Thomas cluster process [105] (which is useful for modelling content placement) [106].

Caching is a typical practical use-case of relay-assisted D2D communication which has the potential of improving the network performance. More research findings would be required to reveal the impact of different realistic assumptions for channel fading and constraints since caching is one of the potential applications of D2D communication.

The amalgamation of cognitive radio with D2D communication has a number of prospects. For instance, the use of cognitive D2D reduces the challenge of allocating frequencies by the operator. Furthermore, proper interference management and resource utilization can be achieved [22]. Looking into existing work on the TC of cognitive radio systems, we can take a few lessons such as the inclusion of parameters like medium access probability as used in [107]. For instance, the achievable TC can be studied under diverse medium access schemes as presented in [108], while taking the outage constraints of both cellular and D2D systems into consideration. Another lesson is the investigation of TC considering other forms of fading such as block fading channels.<sup>5</sup> The TC can be maximized with constraints in parameters such as transmit power and density.

With regards to TC in other spectrum sharing systems using SG, (1) the trade-offs between the TC of cellular and

115034

D2D systems with different transmission range and power can be investigated for energy harvesting relay-assisted D2D (see [110] for an example in overlaid wireless adhoc networks). Recently, more literature have appeared for realistic non-linear energy harvesting models deployable in this context. (2) The impact of parameters such as mutual distance between transmitters and receivers, BS density and successive interference cancellation can be investigated (see [67] for the case of cellular and mobile adhoc networks) in the context of relay-assisted D2D communication. (3) For cellular networks having guard zones to protect their QoS (e.g., [111]), relays could be used outside these zones to facilitate D2D communication. It is also possible to consider cellular users at the edge of these zones to relay data for other D2D devices. Secrecy performance in mmWave-based relay-assisted D2D networks with blockages is another future direction.

Note that a very common assumption in SG literature is the consideration of specific physical layer configurations such as a particular path-loss condition (e.g.,  $\alpha = 4$ ) and assumption of fixed distances between the transmit-receive pairs. In reality, such assumptions are restrictive [112] and do not provide comprehensive analytical insights into the performance of diverse real-world scenarios. Asides the random relay selection which is quite common in literature, other applicable relay forwarding schemes could be motivated. Recently, the mean distance between the nearest and farthest RNs to the source was derived in the context of cognitive wireless sensor networks (see the Appendix of [113]).

### IV. SPECTRAL EFFICIENCY ANALYSIS AND OTHER METRICS

Stochastic geometry analyses have proven successful in giving analytically tractable expressions and the relationship between parameters which affect the cellular network performance. Asides the capacity analysis discussed in the previous section, other SG-based performance analyses such as SE, average transmission rate and power control for relay-assisted D2D communication are discussed in this section. The lessons learned and issues relating to energy-efficiency are also discussed. A highlight of some key aspects of the reviewed articles is presented in Table 6.

### A. SPECTRAL EFFICIENCY ANALYSIS OF RELAY-ASSISTED D2D COMMUNICATION

In this subsection, we discuss some of the works that studied SE in the context of relay-assisted D2D communication using SG and describe a typical spectrum sharing model for the articles reviewed.

#### B. REFERENCE MODEL

In this model, the BS is modelled as a PPP with intensity  $\lambda_{BS}$ . D2D users are modelled as a PPP ( $\Phi_d$ ) thinned from the universal set ( $\Phi_u$  with intensity  $\lambda_u$ ) which includes cellular and D2D users. The UEs could either be active (belonging to  $\Phi_u$  with intensity  $\lambda_u$ ) or inactive (belonging to  $\tilde{\Phi}_u$  with

<sup>&</sup>lt;sup>5</sup>In cognitive radio systems, [109] studied the outage capacity and ergodic capacity of secondary users in a block fading channel environment.

	Aspects	К		References				
	rispects		[89]	[4]	[91]	[90]	[92]	[93]
ion	Derivative-based algorithm	✓						
izat	Ergodic capacity power-aware optimization		$\checkmark$					
tim	Power control optimization				✓			
l o	Energy efficiency maximization	<b>√</b>		$\checkmark$	<ul> <li>✓</li> </ul>			
	Closest RN to transmitter selection					<ul> <li>✓</li> </ul>	$\checkmark$	
ons	Relaying through idle-users						$\checkmark$	
] Ipti	Partial decode-and-forward						$\checkmark$	
Assun	Power control			$\checkmark$	<ul> <li>✓</li> </ul>			
	Out-of-cell interference consideration						$\checkmark$	
	Frequency Division Duplexing							$\checkmark$
	Coverage extension							$\checkmark$
ives	Spectral-efficiency analysis					<ul> <li>✓</li> </ul>		
ject	Energy-efficiency analysis	$\checkmark$		$\checkmark$				
g g	Protect cellular transmission					$\checkmark$		
	Evaluation of Average transmission rate						$\checkmark$	
ಟ್ಟ	Evaluation using 3GPP parameters							$\checkmark$
lelir	Interference modelling using Gamma distribution						$\checkmark$	
4od	Mission-critical application							$\checkmark$
~	Poisson point process	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
	Rician fading		$\checkmark$					
ts	Layer 3 relays				$\checkmark$			
pec	Spatially correlated interference					$\checkmark$		
As	Monte-Carlo simulation			$\checkmark$	$\checkmark$			
dne	Interference suppression	<ul> <li>✓</li> </ul>			$\checkmark$			
Dii	Spectrum partition factor					<ul> <li>✓</li> </ul>		
_	Closed-form expressions		$\checkmark$	$\checkmark$				
	Energy savings via Relays			$\checkmark$				
	Ergodic capacity vs. power trade-off		$\checkmark$					

TABLE 6. Key aspects of the reviewed literature on other performance analyses of relay-assisted D2D communication.

intensity  $\tilde{\lambda}_u$ ). For simplicity of representation, assume that a tagged BS covers an area approximated by a circular disk with radius  $A(k, R_{BS})$  which is based on the approximation on a disk of radius  $R_{BS}$  centered at the typical BS k. The Bernoulli random variable model is shown in Eq. (4) below.

$$\Pr\left(\sim \mathbf{a}\right) = \frac{\tilde{\lambda}_{u}}{\lambda_{u} + \tilde{\lambda}_{u}}.$$
(4)

Note that the intensity of relay users is  $\lambda_R = \lambda_u$  and it is implicitly assumed that inactive users can function as relay nodes for D2D communication. The topology in Fig. 10 is assumed to be semi-static i.e., remains the same for a long period of time. Given that the set of D2D devices is obtained from thinning PPP for the total number of users, the intensity of D2D users is thus  $\lambda_d = q\lambda_u$ , where q = $\Pr(N_r = 0)\Pr(SINR_{t,d} \ge \theta_D) + \Pr(N_r \ge 1)\Pr(SINR_{t,r} \ge \theta_D)$ . SINR<sub>t,d</sub>, SINR<sub>t,r</sub> and SINR<sub>r,d</sub> represent the SINR threshold between the transmitting node and destination node, source node and RN, and RN and destination node, respectively [90]. All devices can be assumed to have a constant transmit power or channel inversion power control could be used. This model has been used in [90] and a similar model was used in [114].

#### C. RELATED LITERATURE

#### 1) SPATIO-TEMPORAL INTERFERENCE CORRELATION

Temporal and spatial correlations occur in a wireless system where the transmitting set changes at every time slot (due to the medium access scheme/scheduler). The transmitting nodes at different time slots are usually chosen from a set having the same source of randomness [115]. The phenomenon where interference is correlated may make the performance of wireless communication systems degrade significantly in a spatial and temporal manner [116]. Interference correlation is either studied on the basis of static network or networks with finite mobility. Specifically, as the level of mobility tends to infinity, the interference is decorrelated across space and time [117]. Thus in real-world settings where topology changes occur less frequently as compared to packet retransmission and fading fluctuations, spatio-temporal correlation comes into the picture as the performance of active links are location dependent. As a result, the SIR seen by close receivers leads to spatially correlated SIRs [118].

In this respect, Atat *et al.* [90] study the spatial-temporal correlation between interference at the RN and destination device in relay-assisted D2D networks using the reference model in Fig. 10. Considering the correlation of aggregate interference at RNs (see Fig. 11), they investigated the SE



**FIGURE 10.** A reference model with relay nodes distributed within angle  $\bigcirc = \left[ -\frac{\pi}{2}, \frac{\pi}{2} \right]$  [90] © IEEE.



**FIGURE 11.** Network model showing spatio-temporal correlation of interference between communication links in a relay-assisted D2D network [90] © IEEE.

of relay-assisted D2D communication. A tractable analytical framework was presented to statistically analyze the performance of relay-assisted D2D networks. This accounts for the SE characterization of both cellular and D2D links. The framework gives insights into the achievable spectral efficiency gains when relays are deployed.

#### 2) POWER CONTROL

This is crucial for the efficient use of energy and coordinating interference in wireless network [3]. It is essential for attaining the target SINR of the receiver. Also, the transmit range of devices depends on the transmission power which impacts the network throughput and connectivity. When transmit power of devices are reduced, the interference experienced by other 'ongoing' transmissions can be reduced. Therefore, this promotes more concurrent transmissions within the network [119]. Power control is a well-studied approach for managing the interference when D2D devices coexist with cellular devices. This helps to ensure that they do not interfere with traditional cellular communication [82].

On this subject, [91] proposed a power control scheme aimed at mitigating and suppressing interference between D2D and cellular communication to improve the SE and throughput performance. For simplicity, they considered a



FIGURE 12. Network model showing fixed layer 3 relays [91] © IEEE.

single-cell scenario having three layer-3 RNs with fixed locations as shown in Fig. 12. In this model, one cellular user and D2D transmitters simultaneously share the resources in the uplink. The RNs transmitting to D2D receivers utilize the same uplink resources. Using the relay assistance, the authors showed an improvement in the coverage probability of the cellular link when devices are far apart. The research examines cellular users that are uniformly located within the coverage region of the BS. Future work can study the use of such fixed layer-3 relaying scheme in a multi-cell environment which takes interference of other cells into account. However, this has to be motivated by applicable real-world scenarios since specifying a fixed number of relays appears restrictive and may need to be generalized. Furthermore, the spatiotemporal correlation of interference can be considered at the relays and receiving devices.

#### D. TRANSMISSION RATE AND SERVICE SUCCESS PROBABILITY

This subsection reviews SG research on metrics such as coverage, throughput and transmission rate, in the context of relay-assisted D2D communication. We note that some of the articles reviewed here also studied other metrics.

#### 1) TRANSMISSION RATE

A pioneering effort that analyzed user-assisted relaying in network-wide context is [92]. The authors studied partial decode and forward user-assisted relaying in the cellular uplink. In the scenario investigated, an active user relays data through the nearest idle user whenever D2D communication is enabled between two UEs in proximity. The proposed cooperation policies cater for fast, slow and hybrid fading channels. Analytical results were compared with the ideal situation when nodes have perfect information about the channel state. Their findings indicate that per-user transmission rate can be improved significantly using user-assisted relaying regardless of the out-of-cell interference. Transmission rate gain was also shown to increase with higher idle UE density and more preponderant in the case when active users are closer to the cell edge.

#### 2) SERVICE SUCCESS PROBABILITY

The authors in [93] quantify the cellular network performance in terms of service success probability during network infrastructure breakdown such as when a natural disaster occurs. They modelled the link-level cellular service probability considering both successful uplink and downlink performances as  $P_c = P_{UL}P_{DL}$ , where  $P_{UL}$  and  $P_{DL}$  are the uplink and downlink success probabilities, respectively. The network level-performance of a single-hop relay network was derived as  $\varrho_1 = \varrho_o + \varrho_1 - \varrho_o \varrho_1$ , where  $\varrho_1$  is the performance of the one-hop relay link (without the direct link) and  $\varrho_o$  is the average link-level success probability for the cellular link. For an *n*th relaying chain, the network-level performance  $\varrho_n$  is related to the the (n - 1)th relaying chain network-level performance'  $Q_n$  is as shown in Eq. (5).

$$\varrho_n = \varrho_{n-1} + \varrho_n - \varrho_{n-1} \varrho_n 
= \varrho_{n-1} + \varrho_o \varrho_o^n - \varrho_{n-1} \varrho_o \varrho_o^n.$$
(5)

After disaster occurrence, the BS was modelled with a thinning factor having a new resulting density  $(1 - \eta)\lambda_{BS}$ , where  $\eta \in [0, 1]$  is a proposed damage ratio which indicates the percentage of 'phased out BS' with respect to the total number of BS. Analytically, the authors evaluated the effect of D2D relaying to alleviate damages in such scenarios as  $\rho_n - \rho_0$ . Attempts were made to answer the following questions.

- How much alleviation can be achieved when D2D technology is introduced to the disaster scenarios?
- How can the network-level service success probability be described when chain relaying is used to extend cellular coverage?
- What insights can be deduced when disaster propagation scenarios are considered?

Results show that a buffer of 50-70% can be attained when a single hop relay is deployed for D2D communication.

#### E. LESSONS LEARNED

Spatial and temporal correlation of interference is a fundamental aspect of wireless communication scenarios with static, quasi-static or finite mobility. However, the performance analysis of relay-assisted D2D which considers spatio-temporal interference correlation is quite recent. This can be included in the fine-grained SG-based analysis of network models incorporating D2D communication in future works. From this review, we observed that analytical methods are mainly used within this domain to study network performance. Monte-Carlo simulations are also used (in some cases) to verify the correctness of the analytical results.

One of the key peculiarities of D2D communication is its applicability in critical scenarios where traditional cellular communication may not be effective, such as the event of natural disasters. SG can be deployed to provide insights into the network performance and relay link gains for such a scenario. It is worth noting that UAVs can also facilitate

VOLUME 7, 2019

communication among devices during disasters. In addition to the aforementioned, power control is essential to meet the QoS requirement of the cellular network. In this respect, SG models have appeared in D2D and relay-assisted D2D literature to study this phenomenon in a tractable manner.

Similar to some works on TC discussed in Section III, optimization can be used with SG to evaluate the energy-efficiency of D2D communication. This tool has been used to study energy efficiency in multiple bands and reduce the overall network power consumption (see [97]). From our survey, we note that the Rayleigh assumption is the most common assumption used in SG-based investigations of relay-assisted D2D in the literature. A similar observation was made by Elsawy *et al.* [16]. The authors noted that this is due to the tractability and simplicity associated with the Rayleigh fading assumption.

Looking into the SG-based analysis in other wireless systems, several interference distributions can be analyzed (see [120]) to determine the desirable characteristics of spectrum sharing systems involving cellular, D2D, and relay nodes. Comparisons of the capacity and capacity gain of the relay-assisted D2D network can be made under various fading distributions similar to [121]. Fairness coexistence constraint [58] could be studied in the context of relay-assisted D2D communication considering the optimal transmit power and spatial densities in such scenarios. UAVs can also be incorporated into such models. However, this would require the consideration of a 3D model.

In recent years mobile technology has evolved from the usual call and text messages services to multimedia sharing. To improve the SE of cellular networks and facilitate multimedia sharing, D2D communication has proven to be very useful [122]. The current multimedia content-sharing applications consume battery. Also, the coexistence of D2D devices brings about interference to the cellular system [3], [97]. These indicate the need for energy-efficient D2D communication. The use of RNs to assist D2D communications can improve the energy-efficiency. However, the trade-offs between the transmit powers of D2D and RNs should be investigated in diverse network scenarios and application use-cases.

The consideration of inter-cell interference in relayassisted D2D-enabled networks is another area for future work. Typical related work in this regard for Frequency division multiple access (FDMA) cellular network assisted by relays is given in [123]. It is important to bear in mind that the study of relay-assisted D2D communication for mmWave networks is still an emerging research direction. Future work could also study the impact of power control on the outage probability of cellular systems in the context of relay-assisted D2D in mmWave bands. Furthermore, the benefit of using relays (mobile and static) especially at the cell edge to improve the quality of cellular communication is worth giving further consideration.

Another aspect that has received much attention in SG-based research for relay-assisted wireless systems is relay

selection [124]–[130]. In this regard, several relay selection schemes have been proposed. Also, attention has been given to the use of mobile relays (see [84], [128], [131]) which include both terrestrial and airborne Unmanned Aerial Vehicles (UAVs). Opportunities include proposing architectures for several use cases such as disaster management and crowdsourcing (e.g., how UAV-assisted D2D communication can be used in such scenarios). In a multihop scenario, if a typical receiving device is not within the range of the sending device, a nearby UAV can receive the data and send to the most appropriate device closer to a destination device towards the control center for appropriate actions and rescue operations. These relays could also be equipped with energy harvesting capabilities. Also, optimal relay selection algorithms for these scenarios would be sought after in such research. The impact of various relay positioning should also be investigated.

## V. FURTHER DIRECTIONS ON RELAY-ASSISTED D2D COMMUNICATION

Following the review above, we present future directions related to the study of relay-assisted D2D communication using SG tools as shown in Fig. 13. Some of these directions could also be jointly investigated for D2D communications.

#### A. RELAY-ASSISTED D2D FOR CACHING

One of the key benefits of D2D communication is its potential to offload traffic from the BS. Particularly, this becomes very important when users are demanding for popular files such as multimedia content. In such cases, the traditional BS may not be able to cater for such overwhelming demand. Caching files at a nearer device is one of the key solutions to this problem. Devices could easily retrieve cached files from another device via D2D links. Besides alleviating the traffic load at the backhaul, this has the potential to reduce the delay in the delivery of such files and facilitate quick access to multimedia content [132]–[134].

An interesting pattern in our age is the rate at which popular content gets viewed, disseminated and downloaded. In YouTube, for example, a published video can have one million views and thousands of downloads within a short period of time. Imagine the amount of traffic that would be directed at such content. To handle such traffic bursts, a bright idea is to cache those files within the reach of users so that they can have it downloaded or viewed without placing any burden on the backhaul. However, D2D devices also have their constraint in terms of memory and the distance at which they can be effectively functional. This brings in the use of relays [132], to retrieve cached files from nearby users. Note that several objectives of interest in cache-enabled D2D communication relate to optimizing content placement [135]. In this context, relays can be deployed to manage this problem by facilitating communication between the requester of content and the devices in which they are cached.

Using network coding-based caching policies improves cellular network performance when compared with traditional caching techniques. This is because a linear combination of file chunks can be stored in the cache of the BS which yields a more effective performance [136]. It is also imperative to mention that cellular and D2D connections can cooperatively be used to retrieve missing content as a result of packet loss using network coding [137]. Moreover, energy harvesting relays can be used to achieve cooperative communication in a two-way network-coded network as studied in [138]. The approach was not considered for content caching, therefore, it opens opportunities for network coding-based relay-assisted cache-enabled D2D communication. Network coding can also be used to guarantee successful delivery probability using broadcast [139] that is useful in the context of sharing cached files requested by a group of users. Note that the above works ([136]-[139]) are typical examples of the deployment of SG tools to study the network coding phenomenon in diverse scenarios (among which is caching).

The use of SG tools to study relay-assisted caching and content placement is an emerging area with few published findings. It brings many forms of point processes into consideration since caching architecture depends on the node distributions within the application area. Furthermore, selecting the appropriate RN to achieve reduced delay and better performance can be fused into proposed frameworks in this domain. Proper mode selection for DUEs and CUEs should also be considered to satisfy the requests of devices requiring cached files. Summarily, SG can provide analytical tools robust to give insights into several questions related to D2D and relay-assisted D2D caching in 5G wireless networks.

#### B. TWO-WAY RELAYING

Two-way relaying is another promising direction which has been hardly explored in the context of D2D communication using SG. In two-way relay channels, nodes can exchange data concurrently using one or more relays [140]. The benefit of this is that it can achieve two-fold SE [141] and the communication is bidirectional i.e., as the source device sends data to the destination device, the destination device has data to send to the source device [140], [141]. This assumption is practical since it can be found in uplink and downlink cellular communication as well as packet acknowledgment [140]. Thus it is considerable in D2D communication especially when control messages would have to be sent between devices using relays.

The drawback of this approach is managing selfinterference, synchronizing source nodes, and channel estimation [141]. Existing research on traditional two-way relaying include [142], [143]. It is important to bear in mind that when the transmitters are PPP distributed, the bidirectional interference becomes correlated and this is quite challenging to compute [144]. In recent times, some analytical efforts such as [90] have been directed towards spatio-temporal interference correlation in D2D networks. Two-way relaying can also be studied in the context of energy harvesting (see [145]–[152]) and wireless power transfer (see [153]–[157]).



FIGURE 13. Organization of the further directions on relay-assisted D2D communication presented in this paper.

#### C. INTERFERENCE CANCELLATION

One of the main limitations of the cellular network performance is interference caused by 'human designed' devices [158]. When DUEs and relays share the spectrum with cellular users, this interference further aggravates. Similarly, in a scenario with dense network infrastructure there exist an advantage of improved signal power (and signal strength) which comes at the cost of an almost equal increase in interference, thereby limiting the SE gains. Two major techniques used to tackle interference are interference avoidance/coordination and interference cancellation. The former may not be as competitive as the latter when spectrum utilization is taken to account. This is because interference cancellation facilitates an aggressive spectrum reuse [159]. Thus, IC is a typical approach to mitigating interference in wireless networks where the desired information is demodulated and (or) decoded. The interference from the received signal is then canceled using this information in conjunction with channel estimates. A challenge, however, is providing accurate models to reconstruct the signals as well as model what such signals had 'experienced' when passing through the channel [158].

Parallel and successive IC are the practical techniques of IC [158]. To the best of our knowledge, no research has modelled parallel interference cancellation using SG. Also, IC is yet to be studied for relay-assisted D2D communication using SG, let alone a scenario where some devices can harvest energy and others can cancel interference.<sup>6</sup> SG tools would help present further research insights in this domain since it has reached an appreciable level of maturity in deriving both closed-form expression as well as tight upper and lower bounds on key metrics in the network. IC can also be studied with respect to content caching for diverse caching policies. In such scenarios, interference cancellation coefficients (see [62]) may be infused into the analysis. Additionally, several technologies such as machine-to-machine communications could co-exist with relay-assisted D2D communication. This would require network performance characterization and proper IC measures.

#### D. RELAYING WITH PACKET RETRANSMISSIONS

The inclusion of packet retransmission into SG-based D2D underlay cellular networks has been considered by Cardieri and Brito [160]. This opens a promising direction since the assumption of packet retransmission can improve the performance of the network.

There is no doubt that when relays are used to transfer data from the source to destination device and packets are retransmitted, more interference exists within the network. Managing such interference while enjoying the benefits of packet retransmission is thus an open issue. A key parameter in this regard is the packet arrival probability as deployed in [160]. The TC expression in such scenarios could also be determined. Such models can be studied in the context of energy-harvesting relays.

#### E. SOCIALLY-AWARE RELAYING

From our survey, we discovered that only [161] studied socially-aware relay-assisted D2D communication using SG. A recent research in M2M communication ([162]) analyzed the *k*th community. This paradigm (i.e., socially aware D2D) brings the realistic involvement of people and mobile devices into picture [163] and thus could depict a more practical scenario of D2D communication which involves human-held devices. Furthermore, socially aware-D2D communication can be used to achieve improved system performance [164].

From a social perspective, not all devices would be willing to function as relays. For instance, a D2D user may decide not to relay data for another user when they are strangers to each other. Although the use of incentives has been studied recently, it is more natural that users are willing to relay data for other users when they both exist in the same physical and social domain (e.g., students' acquaintances in a classroom). Amongst others, social attributes include users

<sup>6</sup>Note that in most cases, the energy harvesting efficiency is less than one.

mutual interest and background [165]. In this respect, social ties can be used to improve the performance of cooperative D2D communication [166]. These should be considered in the investigation of relay-assisted D2D communication.

Asides the aforementioned, the consideration of trust which has recently attracted researchers interest [167] is a potential direction in socially-aware relay-assisted D2D communication. Trusts could be assigned values [168] where the willingness to cooperate is modelled as a value between 0 and 1. In this case, a trust level of 0 implies distrust and reluctance to cooperate while 1 is an indication of a high level of trust and willingness to cooperate. Another recently studied direction with regards to trust-aware relay selection is the consideration of a buffer-aided scheme [167], where relays can store data in buffers. Transmission of the stored data can be deferred to the time when the links are significantly strong. The consideration of buffers is a promising direction in the analysis of SG-based relay-assisted D2D communication because it combines queuing theory and SG which researchers have found very beneficial. However, it is important to note that packet queuing delays should be kept minimal. This requires that buffers should be maintained for a very short time [167]. Interference management within such framework and metrics such as energy-efficiency can be studied in future works.

#### F. UAV-ASSISTED D2D COMMUNICATION

The deployment of UAVs for information transfer has become one of the most recent research directions in wireless communications. This is because UAVs bring a new dimension to the existing cellular communication framework where flying drones can increase the coverage and capacity of the network. Specifically, the cellular infrastructure benefits from their flexibility, mobility and adaptive altitude. These make this technology highly propitious for public safety and emergency applications, especially when the terrestrial network is not fully operational or experiences a damage [169]. Asides its potential to provide connectivity, this solution could also be cost-effective [170] in critical applications and can also be 'swiftly-deployed' [171]. Besides, flying drones could be used to offload the traffic at the BS [170].

In an emergency response application (where D2D is helpful), UAV-relaying might provide connectivity between users (or groups) distant from each other especially when these users/groups cannot effectively communicate directly [171]. Hotspots, such as a university campus, can benefit improved throughput through UAVs [172]. Very recently, SG-based UAV models have begun to receive research attention [173]–[175]. Specifically for D2D communication, the coexistence of UAVs will bring in many challenges with regards to interference management in various UAV trajectories. Developing optimal UAV trajectories to improve energy efficiency as well as analyzing TC of UAV-assisted D2D communication are areas for future consideration.

Quite possibly, UAVs may deputize a malfunctioning BS or assist a macrocell when the radio access network or core

network is congested [176]. The peculiarities of clustered D2D served by UAVs hovering above cluster centers should be taken into account. For example, instead of considering static clusters [177], dynamic clusters having mobility can be studied. Also, for a multiple UAV scenario, UAV height may be assumed to follow a particular distribution as compared to having equal height. UAV-assisted D2D with respect to mmWave and sub-6GHz band are conceivable. In the course of modelling terrestrial communication, a more generalized fading channel may be used as compared to the common Rayleigh fading assumption. Rigorous downlink and uplink analysis can be studied with respect to power control, optimal height and energy harvesting UAV-assisted D2D where DUEs harvest from UAVs. Non-linear energy harvesting models are considerable in such scenarios.

#### G. ENERGY HARVESTING RELAYS

A promising solution for the future highly dense heterogeneous networks is wireless powered communication. This paradigm has diverse practical applications [178]. Looking into the prospects and expectations of future D2D applications, we can see that D2D devices would consume more energy during their course of operation. Although mobile users currently charge their devices, the increased workload on these devices would require alternative energy sources to complement their batteries. Therefore, it is imperative to ensure that relays are equipped with an alternative energy source. Otherwise, users would feel reluctant to relay data for others (except when there are social ties or strong incentives). One way to achieve energy-efficiency and increased throughput is energy harvesting. The objective is to prolong the battery life of devices while harvesting from sources like solar, thermal, wind, ambient RF signals, etc.

The opportunities in this area using SG include proposing architectures involving any of the socially-aware characteristics, caching files at energy-harvesting relays, UAV-assisted file transmission (such as broadcasting a file among devices with common interest within a geographic region), etc. Furthermore, metrics such as outage probability, SE and TC could all be studied using the framework. As energy harvesting relays can harvest energy from access points (see [179] for example), future work can present SG models studying the scenario where D2D devices harvest from a combination of any of cellular devices, IoT devices and UAVs. This requires the derivation of the associated energy harvesting success probability<sup>7</sup> and appropriate point processes for these devices. Particularly, a 3D point process would be required for flying UAVs. In addition to the trade-offs between energy harvesting and transmission time, the impact of channel inversion power control and spatio-temporal correlation of interference should be given due attention.

Note that many of the works (such as [179]–[181]) that studied relay-assisted communication in heterogeneous

networks and cognitive radio using SG have deployed linear energy harvesting model as compared to the model with a 'variable conversion efficiency [182]'. The latter is more realistic and was used by Mekikis *et al.* [112] in the context of wireless sensor networks. Recently, more models have appeared in literature which can be adapted to relay-assisted D2D communication. Specifically, curve fitting could be used for existing measurement results on hardware circuits for energy-harvesting which match the specification of a typical D2D device. This can be adopted to the network analysis in contrast to the linear model.

#### H. RELAY-ASSISTED MILLIMETER-WAVE

The global shortage of bandwidth for mobile broadband data access makes mmWave communication a promising technology for future generation cellular networks. Millimeter-wave is attractive considering its high bandwidth and reduced antenna size which enable a large number of antennas within a small space. These bring prospects for technologies such as massive MIMO [183]. However, mmWave has poor diffraction potential and is susceptible to blockages due to penetration loss. As a result, mmWave mainly relies on lineof-sight (LOS) transmission which makes high path loss, atmospheric absorption, high noise power and sensitivity typical challenges of mmWave communication. These factors impair its signals and thus lead to inadequate network coverage [183]–[185]. Note that although mmWave helps to 'achieve' a lower interference due to directionality, there would be degradation in the desired signal quality due to blockage [183]. When there is no blockage, transmission can occur directly between source and destination nodes without the help of relays [186].

To reap the benefits of mmWave technology while curtailing its major challenges, the use of relays is highly propitious [184]. Research in this area is quite sparse and only very few investigations have been made using SG. Perhaps, the first systematic performance evaluation on D2D relay-assisted mmWave cellular communication is presented in [184] where the authors study a dual hop D2D relaying scheme and develop a model for the coverage probability in the downlink. In this context, further SG research could account for blockages experienced in diverse scenarios with interfering nodes, beamforming, LOS and non-LOS propagations. The gains in coverage, capacity, and SE can be considered in such cases. Furthermore, studied models could put socially-awareness and ambient RF harvesting into consideration.

Transmit power and distance-based path loss that affect mmWave-based network performance can be studied for relay-assisted mmWave-based D2D communication. mmWave-based cellular networks can further improve the performance of cache-enabled D2D communication due to the high bandwidth availability [187]. This is very promising for sharing very large files since the constraints in bandwidth is less compared to using microwave frequencies. The use of relays would further help to improve coverage in such

 $<sup>^{7}</sup>$ Energy threshold beyond which devices can relay data after harvesting sufficient energy.

scenarios when contents are to be requested from devices which are far away, or when there is a blockage that prevents proper LOS transmission.

The secrecy capacity of social-aware mmWave-based energy harvesting relay-assisted D2D communication is another future direction which can be studied for different spatial node configuration motivated by suitable application scenarios. Existing analysis on secrecy in the context of D2D communication using SG [188]–[190] was not performed in the context of relay-assisted D2D communication let alone with respect to mmWave-based frequency bands and social awareness.

#### I. GENERALIZED FADING CHANNELS

D2D communication has become a key component of 5G networks by enabling the mutual connection between devices. Similarly, it helps to improve the network throughput, however, this comes at the cost of interference with cellular and other D2D users. This is because the physical channel is affected by factors which impact the communication quality (for instance, the propagation characteristics of the physical channel vary and as such, conventional fading models may not be sufficient for modelling) [191]. The operation of 5G will create various link requirements and the need for wireless devices to function over a wide variety of channels ranging from LOS to non-LOS, indoor to outdoor, homogeneous diffuse scattering to those with clustering of scattered multipath waves. Most works on SG-analysis for wireless networks focused on Rayleigh fading distribution as the typical small-scale fading model due to its simplicity [161]. However, Rayleigh fading distribution is very particular and thus may not capture new real-world fading environments with a wide variety of usage scenarios and link requirements [192].

Several models have been used to study other distributions involving large-scale shadowing effects, random shadowing and general fading distributions.<sup>8</sup> A recent study of D2D communication under generalized fading channels including Rayleigh, Rice, Hoyt, Nakagami-*m* and one-sided Gaussian models is presented in [191]. However, there is still much to be done with regards to studying relay-assisted D2D communication in terms of TC and other metrics like local delay under generalized fading channels. Similarly, the study of ( $\kappa - \mu$ ) shadowing is considered an attractive option which captures several real-world applications including D2D communication [161]. For other 'non-Rayleigh' fading distributions, closed-form expressions which do not require a complex numerical evaluation are desired [192].

#### J. SPATIO-TEMPORAL CORRELATION

A quick occurrence of retransmission after a transmission failure is likely to fail another time if a certain receiver has a temporal interference correlation. In vehicular networks where broadcast communication occurs and interference is highly spatial-temporally correlated, there could be a high correlation of outage within a dense cluster which leads to poor performance in terms of information dissemination [116]. This should be put into consideration when modelling D2D communication for vehicular networks involving such broadcast messages. Note that in the context of retransmissions and packet forwarding/routing, the spatial and temporal correlation of interference also makes the outage spatio-temporally correlated [115]. Traditional SG approaches aimed at presenting a *first-order* analysis confined to a single point of observation and usually a single-time frequency resource 'slice', are not sufficient to characterize the spatio-temporal correlation of the performance of a randomly chosen user. A joint analysis of observations from two spatial positions/locations is required *and/or* different 'timefrequency resource slices' [117].

Spatio-temporal correlation in cellular networks is challenging since the policy for BS association has to be considered. In such cases, interference cannot be modeled as infinite homogeneous PPP like adhoc networks [117]. The relationship between correlation distance and correlation of interference as well as the impact of node mobility on correlated shadowing environment was just recently reported in [116]. It is important to study the connection between correlation distance of shadowing and the temporal correlation of interference. Only a few research considered spatially correlated shadowing although many have studied the impact of correlation of node location/traffic. Slow fading is spatially correlated on a scale of 50-200m due to blockage and reflection in transmission channels [116]. Recently, the spatio-temporal correlation of interference was studied for the internet of things where queuing theory was merged with SG [193]. Also, [194] studied spatio-temporal correlation of interference for cooperative retransmission in massive machine type communication. The energy efficiency and TC of spectrum sharing environments for D2D underlaying cellular networks with spatially correlated interference are open for analysis.

Interference correlation can be studied with respect to the coexistence of D2D and other enabling technologies for 5G communication networks such as non-orthogonal multiple access (NOMA) [195]. NOMA involves the cooperative transmission and as such opens up room for the consideration of the interference correlation among receivers in the network. However, we recall that, spatio-temporal interference correlation is often neglected in classical SG literature [196]. NOMA has been studied using SG in diverse scenarios and network configurations such as Poisson networks [197], [198], UAVs [199], [200], caching [201], cognitive radio networks [202], half/full duplex [203], multi-cell mmwave [204], and for D2D communication using tools from SG [196], [205], [206]. Recently, [207] studied meta-distribution and spatio-temporal correlation in heterogeneous networks. Although meta-distribution (see Sec. V-K) has been studied in NOMA [198], [208], it is yet to be studied for relay assisted D2D communication to the best of our knowledge. Wireless energy transfer may further enhance such models.

<sup>&</sup>lt;sup>8</sup>See [161] for critical discussion on their strengths and weaknesses.

#### K. SIR META DISTRIBUTION

Meta-distribution [209] is a new research area which deploys SG. It is the derivation of conditional SIR distribution when the location of wireless nodes are given. This metric gives in-depth information about how the success probability in each link within the network is distributed [210]. It gives a higher level of significance than the traditional success probability which averages over the meta distribution of the entire network [209]. Thus questions relating to the distributions of the per-link success probability for each network realization and the fraction of users which can achieve the desired link reliability at a given SIR threshold can be answered [211]. Moreover, moments of meta distribution and variance of conditional success probability can be derived [211].

Not long ago, meta distribution has been used to study some of the aspects where traditional SG models have been deployed such as BS cooperation [212], interference cancellation [159], power control [211] and mmWave-based D2D networks [213]. A new notion of capacity i.e., spatial outage capacity [214] has been developed using the meta distribution. In the same vein, this can be investigated along with mean local delay for relay-assisted D2D communication. Furthermore, energy harvesting models could also be added to provide further insights and enrich such models. Meta distribution of SIR considering a social-aware D2D framework and successive interference cancellation can also be studied in future works.

#### L. CLUSTERED D2D COMMUNICATION

Content sharing is one of the applications of D2D communication where a device can download a file and share with other devices within its proximity. This may not be accurately modeled using the traditional PPP model hence [105] studied the use of PCP for D2D networks. PCP is more realistic when the clustering effect of device location is taken into account [215]. Asides content-centric networks, applications involving clustered networks include C-RANs (where transmitters are connected to the backhaul), mmWave, indoor scenarios and adhoc networks which involves nodes distributed within a finite region. In such scenarios, a device typically has a nearby device to communicate with directly [216].

Further research could investigate PCP-modelled D2D networks for the spectrum sharing TC and EE. Similarly, relays can be deployed within the PCP model where the devices distributing content can use the relays to connect to devices which are farther away within the cluster.<sup>9</sup> Furthermore, to model devices close to each other and depict the spatial separation between D2D and cellular users by using exclusion zones, a hybrid of PCP and Poisson Hole Process (PHP) has been studied in [103]. The energy gains and TC achieved using this model can be further investigated. These models could be studied with power control for D2D communication.

VOLUME 7, 2019

Note that the statistical dissimilarity of finite point process from different locations makes it very challenging to model finite single-cluster networks [216].

#### M. DELAY CONSTRAINT

Delay is a crucial performance metric in delay-sensitive cellular network applications like multimedia and internet-voice applications [217]. It is a major component of mobile traffic streaming services (voice over internet protocol/video conferencing) and elastic traffic services (such as web browsing and file transfer). The former is delay-sensitive but loss-tolerant while the latter is delay-tolerant but loss-sensitive. The third emerging class has a relaxed delay constraint and flexible traffic volume. This applies in proactive caching systems where content is pushed and cached in closer proximity to the end users. This kind of traffic could be referred to as ultra-elastic [218]. Forms of delay in a wireless network include transmission delay (the time spent on successful data transmission) and queuing delay (the waiting time in one or more service queues). A local delay is a form of transmission delay which could give a picture of the lower bound of the overall system delay. When the local delay is infinite, users do not get any useful service [219].

In this direction, [217] recently investigated D2D performance with delay constraint for energy consumption, STP, and TC. Future work could investigate relay-assisted D2D communication with delay constraint. Other point processes such as PCP could also be studied for their delay constraints. Following recent investigations, the interference characterization of D2D transmitters in a clustered framework with delay constraint could be studied [220]. Loss sensitive traffic should be analytically characterized as used in [218]. SG and queuing theory are promising approaches to capture this in a tractable manner [218].

#### N. IN-BAND FULL DUPLEX RELAYING

In the context of wireless networks, the word 'duplex' is meant to imply the ability of two systems to mutually communicate with each other. Duplexing modes are basically half duplex or full duplex. In the former devices cannot transmit and receive simultaneously which makes it easy to implement. It is thus the most commonly used duplexing technique [221]. The latter implies that devices can simultaneously transmit and receive data. This could either be out-band where each mobile/BS transmits and receives data using orthogonal resources or in-band where simultaneous transmission and reception can occur at each wireless terminal over the same frequency band. In theory, in-band full duplex has the potential to double the spectral efficiency of half-duplex. Some other benefits of in-band full duplex are its ability to double the ergodic capacity, reduce feedback and end-to-end delays. However, the challenges it faces include self-interference, imperfect interference cancellation (as self-interference cannot be perfectly canceled), increased inter-user interference, increased consumed power and complexity [222]. Although the existing 4G deploys

<sup>&</sup>lt;sup>9</sup>In such cases direct transmission may be feasible, however, relays might provide better performance gains.

half-duplex TDD and frequency division duplexing, a number of authors (e.g., [222], [223]) have emphasized the feasibility of full-duplex communication for 5G wireless systems.

One might wonder – what are the implications of half-duplex and full-duplex relaying? Half duplex implies an inefficient utilization of system resources since an extra dedicated time/frequency slot is required for relaying. This makes full duplex appealing. However, it is essential to design small-sized devices which can operate on full duplex and also ensure efficient channel modelling and estimation, effective cross-layer/joint resource management and proper interference management. The accumulated performance loss is also lower in full duplex D2D [224]. Asides improved spectral efficiency, D2D devices can leverage the short link communication to exploit full duplex communication i.e., with a lower transmit power, it is possible to alleviate self-interference [225].

To manage self-interference in full duplex relays, it is important to have sufficient separation in terms of frequency resources between transmitters, receivers and relays [141]. Reducing co-channel interference while optimizing the system performance can be achieved through power control, link adaptation, beamforming, and scheduling [225]. There is a possibility of an adaptive scheme in the context of D2D [226] and relay-assisted D2D [225]. UEs in the former (D2D without relays) or RNs in the latter (relay-assisted D2D) could choose between full duplex and half duplex modes. However, this requires proper mode selection schemes for devices (or relays) to choose when it is best to function in half or full duplex modes. There is much to be done when stochastic geometry comes into the picture.

Even as researchers like [224], [227], [228] have presented models studying throughput and interference in full-duplex D2D communication, efforts are required to investigate the mode selection using full duplex relays. Further questions in this regard are: when is it most suitable to use full duplex relays? At what distance threshold and transmit power (either fixed or power control) is it most appropriate for half/full duplex devices to deploy full duplex RNs when considering an acceptable level of interference protection for cellular users? From [225] we could further ask: in multi-tier and clustered scenarios, how can the impact of self-interference be reduced? How can we deploy antenna array and beamforming to improve the ability to isolate self-interference for full duplex relays? All these should be researched putting proper self-interference cancellation in mind. We believe from the maturity of current research, the tools of SG can be used to reveal insights into some of these questions.

#### **VI. SUMMARY AND CONCLUSION**

This paper identifies some of the key analytical models in relay-assisted D2D communication using tools from SG. Related literature on the transmission capacities of D2D and relay-assisted D2D communication were examined and compared based on their unique features and assumptions. Besides, insights were drawn from transmission capacity analysis of other spectrum sharing networks. Furthermore, the impacts of parameters such as transmit power, device density, and link distances were studied. It was observed that some of the existing works in this area have assumed equal transmit power for transmitting devices and a random relay selection mechanism. This leaves room for more investigations using power control and other relay selection mechanisms. Additionally, energy harvesting can be included in such models to power energy-constrained relay nodes.

After discussing the capacity analysis of D2D and relay-assisted D2D, the paper overviews other performance metrics in the SG-framework with application use cases such as disaster management. It further discusses other areas to be explored such as the study of UAV-assisted D2D communication while deriving lessons from other spectrum sharing networks. These include the consideration of fairness coefficient constraint, inter-cell interference, SINR and SIR characterization for relay-assisted D2D communication. Threshold-based scheduling and power control are other areas where SG is applicable in the analysis of D2D-enabled cellular networks. The impact of fading on the network capacity, tight bounds on the SIR distribution, and outage probability can be derived in such scenarios.

Moreover, other point processes applicable to scenarios which cannot be best described using the popular PPP model can be studied. In this respect, some of the derived expressions for the characterization of such models in other wireless communication literature may be deployed. Be informed that research deploying artificial intelligence to derive new point processes is on-going and would likely receive its share of researchers time in the coming years. Finally, this survey insights future work for relay-assisted D2D communication by identifying unexplored and under-explored areas. Some of the most recent trends in SG research such as spatio-temporal correlation, delay constraint and SIR Meta distributions were also highlighted. Other future areas for consideration include the SINR characterization in cognitive-based D2D communication and the study of interference cancellation in the context of relay-assisted D2D communication. This paper is a comprehensive eye-opener into the subject of relay-assisted D2D communication with exposition into its diverse opportunities based on recent developments in wireless networks, D2D communication, and SG research.

#### APPENDIX

### USEFUL LEMMAS FOR RELAY-ASSISTED D2D ANALYSIS

### A. LEMMA 1

For a typical receiver in systems  $S_n$ ,  $n \in \{c, d, r\}$ , the general form of the STP should satisfy

$$\Pr\left(\operatorname{SIR}_{n} \geq \theta_{n}\right) = e^{-C_{\alpha}\theta_{n}^{\frac{2}{\eta}}d_{n0}^{2}\sum_{j \in \{c,d,r\}}\lambda_{j}\left(\frac{P_{j}}{P_{n}}\right)^{\frac{2}{\eta}}},\qquad(6)$$

where  $C_{\alpha} = \pi \Gamma(1 + \frac{2}{\eta})\Gamma(1 - \frac{2}{\eta}).$ 

The SIR at the receiver is as shown in Eq. (1) where  $n \in \{c, d, r\}$ . Using the model in Fig. 6, the following



FIGURE 14. Some typical relay selection scenarios.

assumptions should be made before giving further detail on  $d_{n0}$ .

#### 1) NODE DISTRIBUTION

The relay nodes have to be distributed based on a particular underlaying pattern e.g., uniform, cluster, random, etc. For two dimensional networks, node distribution can be in the form of underlaying point process e.g., PPP, Poisson cluster process, Thomas Cluster Process, Binomial point Process, Matern Cluster Process, etc.

#### 2) RELAY NODE SELECTION

There are many characterized relay selection schemes in the literature (see [56], [229]–[232]). These schemes have their trade-offs in terms of spatial reuse, forward progress, and energy-efficiency. Here, we focus on three main characterized selection schemes: the closest relay to the source, random relay selection and farthest relay towards a destination.

When it is assumed that relays with density  $\lambda_r$  are randomly located within a disk centered at a D2D user having radius *R* (the maximum distance allowable between a D2D pair in Fig. 10), with an isotropic direction and Rayleigh distributed distance (i.e., placed according to a 2-D Gaussian distribution [79]), the pdf is given as [90]:

$$f_{R}(r_{x}) = \frac{2\pi\lambda_{r}}{1 - e^{-\pi\lambda_{r}R^{2}}} r_{x}e^{-\lambda_{r}\pi r_{x}^{2}}, \quad 0 \le r_{x} \le R.$$
(7)

This is based on selecting the closest relay to the transmitter. When a relay is randomly and independently placed around the transmitting device in a isotropic manner using a Rayleigh distributed distance without restrictions on the maximum distance, the pdf is given as [79]:

$$f_R(r_x) = 2\pi\lambda_r R^2 r_x e^{-\lambda_r \pi r_x^2}, \quad r_x \ge 0.$$
(8)

Given that  $\odot$  is uniformly distributed between  $\left[\frac{-\pi}{2}, \frac{\pi}{2}\right]$ , the distribution of  $R_z$  in Fig. 10 is given below [90].

$$f_{R_Z}(r_z) = \frac{2r_z}{\pi R^2}, \quad 0 \le r_z \le \sqrt{\pi}R.$$
 (9)

#### 3) DISTANCE DISTRIBUTION

The distance between transmitter and receiver for each hop in a relay-assisted D2D network highly depends on how the relay device is chosen [233] (See Fig. 14). For the forwarding scheme, where the farthest relay to a destination within a transmission range<sup>10</sup> R is selected, the distance between a transmitter and receiver pair is denoted as  $r_{FR}$ . Its distribution is expressed in Eq. (10). Note that this forwarding scheme selects the node which maximizes the progress to the destination device in multihop networks.

$$f_{r_{FR}}(r) = \int_{-\frac{\pi}{2}}^{\frac{\pi}{2}} \frac{\lambda r e^{-\lambda R^2 \left(\cos^{-1}\left[\frac{r}{R}\cos\theta\right] - \frac{r}{R}\cos\theta \sqrt{1 - \left(\frac{r}{R}\cos\theta\right)^2}\right)}}{1 - e^{-\lambda \pi R^2/2}} d\theta,$$
$$0 \le r \le R.$$
(10)

See proof in [231, Appendix A] and background detail in [229, Pg. 40].

As for the nearest relay selection strategy (where the nearest relay in the forward direction is selected to forward data), let the transmit-receive distance be denoted as  $r_{NR}$ . The distance distribution is given in Eq. (11).

$$f_{r_{NR}}(r) = \frac{\lambda \pi r e^{-\lambda \pi r^2}}{1 - e^{-\lambda \pi R^2/2}}, \quad 0 \le r \le R.$$
(11)

For the proof, see [229, Pg. 39] and [231].

For the case when a relay is randomly selected to forward data with the transmit-receive distance  $r_{RND}$ , its distribution is given in Eq. (12).

$$f_{r_{RND}}(r) = \frac{2r}{R^2}, \quad 0 \le r \le R.$$
 (12)

This distribution is the most commonly used in literature.

#### 4) EXPECTED DISTANCE

Thus to define  $d_n$  in our considered scenario, we use the expectation of the relay-link distance. For randomly selected RNs distributed based on PPP in the hatched region of Fig. 6, the expectation is given below [81].

$$E(r) = \tau R, \tag{13}$$

where  $\tau = (4\pi - 36\sqrt{3} + 64) / (12\pi - 9\sqrt{3})$  and E(r) is the expectation of the relay distance.

As mentioned earlier, this expectation is dependent on the distribution of RNs and relay selection scheme chosen. The last component which would impact on the STP of individual links within the network is the duplexing mode as this affects the density of active devices within the network. For TDD, only half of the devices would be active at each point in time.

From Eq. (6), we can see that the parameter  $d_{n0}$  cannot be a fixed value since we are dealing with spatial averages. One way to deal with this is to use the expectation of the distance distribution assumed. This is also associated with the RN selection technique and its distribution function. However,

<sup>&</sup>lt;sup>10</sup>or average transmission range in CSMA-based networks. See [231].

this is just an approximation as a spatial distribution of nodes has been assumed using a point process. Furthermore, the use of an expectation captures the notion of a 'snapshot in time' of point processes.

Going back to Eq. (6) with the assumption of a random relay selection, the probability that a relay node would exist within the intersection region in Fig. 6 based on our PPP assumption is given in Lemma 2.

#### B. LEMMA 2

From Fig. 6, the RN existence probability  $(Pr_e)$  is given as

$$P_{re} = 1 - e^{-\lambda_r (\frac{2\pi}{3} - \frac{\sqrt{3}}{2})R^2},$$
(14)

where the area of the hatched intersection region is based on the area of intersection of two equal circles i.e.,  $\chi = r^2 \left(\frac{2\pi}{3} - \frac{\sqrt{3}}{2}\right)$ . The proof is given in [80], [81]. In TDD network, the density of active users is given as  $\frac{1}{2}\lambda_d P_{re}$  [80], [81]. A more generalized approach to finding the density of active links is given in [230, Equation 12].

#### ACKNOWLEDGMENT

The authors would like to thank everyone who provided valuable support, suggestions, and comments to improve the content, quality, and presentation of this paper.

#### REFERENCES

- P. Mach, Z. Becvar, and T. Vanek, "In-band device-to-device communication in OFDMA cellular networks: A survey and challenges," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 1885–1922, 4th Quart., 2015.
- [2] D. Feng, L. Lu, Y. Yuan-Wu, G. Y. Li, G. Feng, and S. Li, "Deviceto-device communications underlaying cellular networks," *IEEE Trans. Commun.*, vol. 61, no. 8, pp. 3541–3551, Aug. 2013.
- [3] L. Wei, R. Q. Hu, Y. Qian, and G. Wu, "Enable device-to-device communications underlaying cellular networks: Challenges and research aspects," *IEEE Commun. Mag.*, vol. 52, no. 6, pp. 90–96, Jun. 2014.
- [4] A. Al-Hourani, S. Kandeepan, and E. Hossain, "Relay-assisted deviceto-device communication: A stochastic analysis of energy saving," *IEEE Trans. Mobile Comput.*, vol. 15, no. 12, pp. 3129–3141, Dec. 2016.
- [5] H. Chen, L. Liu, T. Novlan, J. D. Matyjas, B. L. Ng, and J. Zhang, "Spatial spectrum sensing-based device-to-device cellular networks," *IEEE Trans. Commun.*, vol. 15, no. 11, pp. 7299–7313, Nov. 2016.
- [6] P. Gandotra, R. K. Jha, and S. Jain, "A survey on device-to-device (D2D) communication: Architecture and security issues," *J. Netw. Comput. Appl.*, vol. 78, pp. 9–29, Jan. 2017.
- [7] H. Feng, H. Wang, X. Chu, and X. Xu, "On the tradeoff between optimal relay selection and protocol design in hybrid D2D networks," in *Proc. IEEE Int. Conf. Commun. Workshop (ICCW)*, Jun. 2015, pp. 705–711.
- [8] X. Luan, J. Wu, C. Piao, Y. Cheng, and H. Xiang, "Cooperative transmission based on multi-relay device-to-device communications in cellular networks," in *Proc. IET Conf.*, Jan. 2015, p. 5.
- [9] V. Wong and C. Leung, "Transmission strategies in multihop mobile packet radio networks," in *Proc. Can. Conf. Elect. Comput. Eng.*, Sep. 1993, pp. 1004–1008.
- [10] Z. Lin, Y. Li, S. Wen, Y. Gao, X. Zhang, and D. Yang, "Stochastic geometry analysis of achievable transmission capacity for relay-assisted device-to-device networks," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2014, pp. 2251–2256.
- [11] H. Elkotby and M. Vu, "Interference and throughput analysis of uplink user-assisted relaying in cellular networks," in *Proc. IEEE 25th Annu. Int. Symp. Pers., Indoor, Mobile Radio Commun. (PIMRC)*, Sep. 2014, pp. 1375–1380.
- [12] M. N. Tehrani, M. Uysal, and H. Yanikomeroglu, "Device-to-device communication in 5G cellular networks: Challenges, solutions, and future directions," *IEEE Commun. Mag.*, vol. 52, no. 5, pp. 86–92, May 2014.

- [13] F. S. Shaikh and R. Wismüller, "Routing in multi-hop cellular device-todevice (D2D) networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 4, pp. 2622–2657, 4th Quart., 2018.
- [14] M. Haenggi, J. G. Andrews, F. Baccelli, O. Dousse, and M. Franceschetti, "Stochastic geometry and random graphs for the analysis and design of wireless networks," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 7, pp. 1029–1046, Sep. 2009.
- [15] J. G. Andrews, R. K. Ganti, M. Haenggi, N. Jindal, and S. Weber, "A primer on spatial modeling and analysis in wireless networks," *IEEE Commun. Mag.*, vol. 48, no. 11, pp. 156–163, Nov. 2010.
- [16] H. Elsawy, E. Hossain, and M. Haenggi, "Stochastic geometry for modeling, analysis, and design of multi-tier and cognitive cellular wireless networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 15, no. 3, pp. 996–1019, Jun. 2013.
- [17] J. G. Andrews, A. K. Gupta, and H. S. Dhillon, "A primer on cellular network analysis using stochastic geometry," 2016, arXiv:1604.03183. [Online]. Available: https://arxiv.org/abs/1604.03183
- [18] H. ElSawy, A. Sultan-Salem, M. S. Alouini, and M. Z. Win, "Modeling and analysis of cellular networks using stochastic geometry: A tutorial," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 1, pp. 167–203, 1st Quart., 2017.
- [19] A. Asadi, Q. Wang, and V. Mancuso, "A survey on device-to-device communication in cellular networks," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 4, pp. 1801–1819, 4th Quart., 2014.
- [20] J. Liu, N. Kato, J. Ma, and N. Kadowaki, "Device-to-device communication in LTE-advanced networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 1923–1940, 4th Quart., 2015.
- [21] F. Jameel, Z. Hamid, F. Jabeen, S. Zeadally, and M. A. Javed, "A survey of device-to-device communications: Research issues and challenges," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 3, pp. 2133–2168, 3rd Quart., 2018.
- [22] P. Gandotra and R. K. Jha, "Device-to-device communication in cellular networks: A survey," J. Netw. Comput. Appl., vol. 71, pp. 99–117, Aug. 2016.
- [23] H. A. U. Mustafa, M. A. Imran, M. Z. Shakir, A. Imran, and R. Tafazolli, "Separation framework: An enabler for cooperative and D2D communication for future 5G networks," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 1, pp. 419–445, 1st Quart., 2016.
- [24] M. Höyhtyä, O. Apilo, and M. Lasanen, "Review of latest advances in 3GPP standardization: D2D communication in 5G systems and its energy consumption models," *Future Internet*, vol. 10, no. 1, p. 3, 2018.
- [25] M. Waqas, Y. Niu, Y. Li, M. Ahmed, D. Jin, S. Chen, and Z. Han, "Mobility-aware device-to-device communications: Principles, practice and challenges," *IEEE Commun. Surveys Tuts.*, to be published. doi: 10.1109/COMST.2019.2923708.
- [26] R. Maallawi, N. Agoulmine, B. Radier, and T. B. Meriem, "A comprehensive survey on offload techniques and management in wireless access and core networks," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 3, pp. 1582–1604, 3rd Quart., 2015.
- [27] U. N. Kar and D. K. Sanyal, "An overview of device-to-device communication in cellular networks," *ICT Express*, vol. 4, no. 4, pp. 203–208, Dec. 2017.
- [28] M. Ahmed, Y. Li, M. Waqas, M. Sheraz, D. Jin, and Z. Han, "A survey on socially aware device-to-device communications," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 3, pp. 2169–2197, 3rd Quart., 2018.
- [29] M. Nitti, G. A. Stelea, V. Popescu, and M. Fadda, "When social networks meet D2D communications: A survey," *Sensors*, vol. 19, no. 2, p. 396, 2019.
- [30] X. Cheng, Y. Li, B. Ai, X. Yin, and Q. Wang, "Device-to-device channel measurements and models: A survey," *IET Commun.*, vol. 9, no. 3, pp. 312–325, 2015.
- [31] C. Kalalas, L. Thrybom, and J. Alonso-Zarate, "Cellular communications for smart grid neighborhood area networks: A survey," *IEEE Access*, vol. 4, pp. 1469–1493, 2016.
- [32] B. Ismaiel, M. Abolhasan, D. Smith, W. Ni, and D. Franklin, "A survey and comparison of device-to-device architecture using LTE unlicensed band," in *Proc. IEEE 85th Veh. Technol. Conf. (VTC Spring)*, Jun. 2017, pp. 1–5.
- [33] G. Misra, A. Agarwal, S. Misra, and K. Agarwal, "Device to device millimeter wave communication in 5G wireless cellular networks (a next generation promising wireless cellular technology)," in *Proc. Int. Conf. Signal Process., Commun., Power Embedded Syst. (SCOPES)*, Oct. 2016, pp. 89–93.

- [34] R. Elouafadi and M. Benjillali, "Cooperative NOMA-based D2D communications: A survey in the 5G/IoT context," in *Proc. 19th IEEE Medit. Electrotechn. Conf. (MELECON)*, May 2018, pp. 132–137.
- [35] M. F. Hashim and N. I. A. Razak, "Ultra-dense networks: Integration with device to device (D2D) communication," *Wireless Pers. Commun.*, vol. 106, no. 2, pp. 911–925, May 2019.
- [36] O. Hayat, R. Ngah, and Y. Zahedi, "In-band device to device (D2D) communication and device discovery: A survey," *Wireless Pers. Commun.*, vol. 106, no. 2, pp. 451–472, May 2019.
- [37] O. A. Amodu, M. Othman, N. K. Noordin, and I. Ahmad, "A primer on design aspects, recent advances, and challenges in cellular deviceto-device communication," *Ad Hoc Netw.*, vol. 94, Nov. 2019, Art. no. 101938.
- [38] M. Haus, M. Waqas, A. Y. Ding, Y. Li, S. Tarkoma, and J. Ott, "Security and privacy in device-to-device (D2D) communication: A review," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 2, pp. 1054–1079, 2nd Quart., 2017.
- [39] M. Wang and Z. Yan, "Security in D2D communications: A review," in *Proc. IEEE Trustcom/BigDataSE/ISPA*, vol. 1, Aug. 2015, pp. 1199–1204.
- [40] M. Wang and Z. Yan, "A survey on security in D2D communications," *Mobile Netw. Appl.*, vol. 22, no. 2, pp. 195–208, Apr. 2017.
- [41] O. N. Hamoud, T. Kenaza, and Y. Challal, "Security in device-todevice communications: A survey," *IET Netw.*, vol. 7, no. 1, pp. 14–22, Jan. 2018.
- [42] R. I. Ansari, C. Chrysostomou, S. A. Hassan, M. Guizani, S. Mumtaz, J. Rodriguez, and J. J. P. C. Rodrigues, "5G D2D networks: Techniques, challenges, and future prospects," *IEEE Syst. J.*, vol. 12, no. 4, pp. 3970– 3984, Dec. 2018.
- [43] S. Ali and A. Ahmad, "Resource allocation, interference management, and mode selection in device-to-device communication: A survey," *Trans. Emerg. Telecommun. Technol.*, vol. 28, no. 7, p. e3148, 2017.
- [44] M. Ahmad, M. Azam, M. Naeem, M. Iqbal, A. Anpalagan, and M. Haneef, "Resource management in D2D communication: An optimization perspective," *J. Netw. Comput. Appl.*, vol. 93, pp. 51–75, Sep. 2017.
- [45] O. Yazdani and G. Mirjalily, "A survey of distributed resource allocation for device-to-device communication in cellular networks," in *Proc. Artif. Intell. Signal Process. Conf. (AISP)*, Oct. 2017, pp. 236–239.
- [46] S. Yu, W. Ejaz, L. Guan, and A. Anpalagan, "Resource allocation schemes in D2D communications: Overview, classification, and challenges," *Wireless Pers. Commun.*, vol. 96, no. 1, pp. 303–322, 2017.
- [47] M. K. Pedhadiya, R. K. Jha, and H. G. Bhatt, "Device to device communication: A survey," J. Netw. Comput. Appl., vol. 129, pp. 71–89, Nov. 2019.
- [48] Y. Li, Z. Kaleem, and K. Chang, "Applications of interference alignment in D2D communications," in *Proc. Int. Conf. Inf. Commun. Technol. Converg. (ICTC)*, Oct. 2015, pp. 993–995.
- [49] S. Umrao, A. Roy, and N. Saxena, "Device-to-device communication from control and frequency perspective: A composite review," *IETE Tech. Rev.*, vol. 34, no. 3, pp. 286–297, 2017.
- [50] O. Bello and S. Zeadally, "Intelligent device-to-device communication in the Internet of Things," *IEEE Syst. J.*, vol. 10, no. 3, pp. 1172–1182, Sep. 2016.
- [51] R. Alkurd, R. M. Shubair, and I. Abualhaol, "Survey on device-to-device communications: Challenges and design issues," in *Proc. IEEE 12th Int. New Circuits Syst. Conf. (NEWCAS)*, Jun. 2014, pp. 361–364.
- [52] P. Li and S. Guo, "Literature survey on cooperative device-todevice communication," in *Cooperative Device-to-Device Communication in Cognitive Radio Cellular Networks* (SpringerBriefs in Computer Science). Cham, Switzerland: Springer, 2014, pp. 7–12.
- [53] K. Shamganth and M. J. N. Sibley, "A survey on relay selection in cooperative device-to-device (D2D) communication for 5G cellular networks," in *Proc. Int. Conf. Energy, Commun., Data Anal. Soft Comput. (ICECDS)*, Aug. 2017, pp. 42–46.
- [54] E. Datsika, A. Antonopoulos, N. Zorba, and C. Verikoukis, "Crossnetwork performance analysis of network coding aided cooperative outband D2D communications," *IEEE Trans. Wireless Commun.*, vol. 16, no. 5, pp. 3176–3188, May 2017.
- [55] S. T. Shah, S. F. Hasan, B.-C. Seet, P. H. J. Chong, and M. Y. Chung, "Device-to-device communications: A contemporary survey," *Wireless Pers. Commun.*, vol. 98, no. 1, pp. 1247–1284, 2017.
  [56] L. Kleinrock and J. Silvester, "Spatial reuse in multihop packet radio
- [56] L. Kleinrock and J. Silvester, "Spatial reuse in multihop packet radio networks," *Proc. IEEE*, vol. 75, no. 1, pp. 156–167, Jan. 1987.

- [57] "Licensed shared access (LSA)," ECC, Allahabad, Inida, ECC Rep. 205, Feb. 2014. [Online]. Available: https://www. ecodocdb.dk/download/baa4087d-e404/ECCREP205.PDF
- [58] J. Lee, J. G. Andrews, and D. Hong, "Spectrum-sharing transmission capacity," *IEEE Trans. Wireless Commun.*, vol. 10, no. 9, pp. 3053–3063, Jul. 2011.
- [59] S. Weber, J. G. Andrews, and N. Jindal, "An overview of the transmission capacity of wireless networks," *IEEE Trans. Commun.*, vol. 58, no. 12, pp. 3593–3604, Dec. 2010.
- [60] S. P. Weber, J. G. Andrews, X. Yang, and G. de Veciana, "Transmission capacity of wireless ad hoc networks with successive interference cancellation," *IEEE Trans. Inf. Theory*, vol. 53, no. 8, pp. 2799–2814, Aug. 2007.
- [61] J. Blomer and N. Jindal, "Transmission capacity of wireless ad hoc networks: Successive interference cancellation vs. joint detection," in *Proc. IEEE Int. Conf. Commun.*, Jun. 2009, pp. 1–5.
- [62] J. Lee, J. G. Andrews, and D. Hong, "Spectrum-sharing transmission capacity with interference cancellation," *IEEE Trans. Commun.*, vol. 61, no. 1, pp. 76–86, Jan. 2013.
- [63] S. Weber, J. G. Andrews, and N. Jindal, "The effect of fading, channel inversion, and threshold scheduling on ad hoc networks," *IEEE Trans. Inf. Theory*, vol. 53, no. 11, pp. 4127–4149, Nov. 2007.
- [64] N. Jindal, S. Weber, and J. G. Andrews, "Fractional power control for decentralized wireless networks," *IEEE Trans. Wireless Commun.*, vol. 7, no. 12, pp. 5482–5492, Dec. 2008.
- [65] N. Jindal, J. G. Andrews, and S. Weber, "Bandwidth partitioning in decentralized wireless networks," *IEEE Trans. Wireless Commun.*, vol. 7, no. 12, pp. 5408–5419, Dec. 2008.
- [66] V. Chandrasekhar and J. G. Andrews, "Spectrum allocation in tiered cellular networks," *IEEE Trans. Commun.*, vol. 57, no. 10, pp. 3059–3068, Oct. 2009.
- [67] K. Huang, V. K. N. Lau, and Y. Chen, "Spectrum sharing between cellular and mobile ad hoc networks: Transmission-capacity trade-off," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 7, pp. 1256–1267, Sep. 2009.
- [68] S. P. Weber, X. Yang, J. G. Andrews, and G. de Veciana, "Transmission capacity of wireless ad hoc networks with outage constraints," *IEEE Trans. Inf. Theory*, vol. 51, no. 12, pp. 4091–4102, Dec. 2005.
- [69] K. Stamatiou, J. G. Proakis, and J. R. Zeidler, "Information efficiency of ad hoc networks with FH-MIMO transceivers," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2007, pp. 3793–3798.
- [70] J. G. Andrews, S. Weber, and M. Haenggi, "Ad hoc networks: To spread or not to spread? [Ad hoc and sensor networks]," *IEEE Commun. Mag.*, vol. 45, no. 12, pp. 84–91, Dec. 2007.
- [71] R. Vaze, "Transmission capacity of spectrum sharing ad hoc networks with multiple antennas," *IEEE Trans. Wireless Commun.*, vol. 10, no. 7, pp. 2334–2340, Jul. 2011.
- [72] R. Vaze and R. W. Heath, Jr., "Transmission capacity of ad-hoc networks with multiple antennas using transmit stream adaptation and interference cancellation," *IEEE Trans. Inf. Theory*, vol. 58, no. 2, pp. 780–792, Feb. 2012.
- [73] J. Ji and W. Chen, "Transmission capacity of two co-existing wireless ad hoc networks with multiple antennas," in *Proc. IEEE Int. Conf. Commun.* (*ICC*), Jun. 2011, pp. 1–6.
- [74] J. Lee, S. Lim, J. G. Andrews, and D. Hong, "Achievable transmission capacity of secondary system in cognitive radio networks," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2010, pp. 1–5.
- [75] M.-S. Alouini and A. J. Goldsmith, "Area spectral efficiency of cellular mobile radio systems," *IEEE Trans. Veh. Technol.*, vol. 48, no. 4, pp. 1047–1066, Jul. 1999.
- [76] A. Kachouh, Y. Nasser, and H. A. Artail, "On the capacity optimization of D2D underlying cellular communications," in *Proc. 23rd Int. Conf. Telecommun. (ICT)*, May 2016, pp. 1–5.
- [77] Z. Liu, T. Peng, H. Chen, and W. Wang, "Transmission capacity of D2D communication under heterogeneous networks with multi-bands," in *Proc. IEEE 77th Veh. Technol. Conf. (VTC Spring)*, Jun. 2013, pp. 1–6.
- [78] Z. Liu, T. Peng, Q. Lu, and W. Wang, "Transmission capacity of D2D communication under heterogeneous networks with dual bands," in *Proc.* 7th Int. ICST Conf. Cogn. Radio Oriented Wireless Netw. Commun. (CROWNCOM), Jun. 2012, pp. 169–174.
- [79] X. Lin, J. G. Andrews, and A. Ghosh, "Spectrum sharing for deviceto-device communication in cellular networks," *IEEE Trans. Wireless Commun.*, vol. 13, no. 12, pp. 6727–6740, Dec. 2014.
- [80] T. Zhou, L. Liu, Y. Zhang, and Y. Yang, "Transmission capacity analysis of relay-assisted device-to-device communication in cellular networks," in *Proc. IEEE/CIC Int. Conf. Commun. China (ICCC)*, Jul. 2016, pp. 1–6.

- [81] Y. Yang, Y. Zhang, L. Dai, J. Li, S. Mumtaz, and J. Rodriguez, "Transmission capacity analysis of relay-assisted device-to-device overlay/underlay communication," *IEEE Trans. Ind. Informat.*, vol. 13, no. 1, pp. 380–389, Feb. 2017.
- [82] S. Wen, X. Zhu, Y. Lin, Z. Lin, X. Zhang, and D. Yang, "Achievable transmission capacity of relay-assisted device-to-device (D2D) communication underlay cellular networks," in *Proc. IEEE 78th Veh. Technol. Conf. (VTC Fall)*, Sep. 2013, pp. 1–5.
- [83] I. Singh and N. P. Singh, "Coverage and capacity analysis of relay-based device-to-device communications underlaid cellular networks," *Eng. Sci. Technol., Int. J.*, vol. 21, no. 5, pp. 834–842, 2018.
- [84] O. Delgado and F. Labeau, "D2D relay selection and fairness on 5G wireless networks," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, Dec. 2016, pp. 1–6.
- [85] Z. Lin, L. Huang, Y. Li, H.-C. Chao, and P. Chen, "Analysis of transmission capacity for multi-mode D2D communication in mobile networks," *Pervasive Mobile Comput.*, vol. 41, pp. 179–191, Oct. 2017.
- [86] L. Fan, Z. Dong, and P. Yuan, "The capacity of device-to-device communication underlaying cellular networks with relay links," *IEEE Access*, vol. 5, pp. 16840–16846, 2017.
- [87] Y. J. Chun, G. B. Colombo, S. L. Cotton, W. G. Scanlon, R. M. Whitaker, and S. M. Allen, "Device-to-device communications: A performance analysis in the context of social comparison-based relaying," *IEEE Trans. Wireless Commun.*, vol. 16, no. 12, pp. 7733–7745, Dec. 2017.
- [88] Y. Yang, Z. Liu, Z. Fu, T. Peng, and W. Wang, "Transmission capacity of device-to-device communication under heterogeneous networks with cellular users assisted," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, Dec. 2013, pp. 341–635.
- [89] C. Liu and B. Natarajan, "Power-aware maximization of ergodic capacity in D2D underlay networks," *IEEE Trans. Veh. Technol.*, vol. 66, no. 3, pp. 2727–2739, Jun. 2017.
- [90] R. Atat, L. Liu, J. Ashdown, M. Medley, and J. Matyjas, "On the performance of relay-assisted D2D networks under spatially correlated interference," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2016, pp. 1–6.
- [91] J.-F. Shi, L. Tao, M. Chen, and Z.-H. Yang, "Power control for relayassisted device-to-device communication underlaying cellular networks," in *Proc. Int. Conf. Wireless Commun. Signal Process. (WCSP)*, Oct. 2015, pp. 1–6.
- [92] H. Elkotby and M. Vu, "Uplink user-assisted relaying in cellular networks," *IEEE Trans. Wireless Commun.*, vol. 14, no. 10, pp. 5468–5483, Oct. 2015.
- [93] A. Al-Hourani, S. Kandeepan, and A. Jamalipour, "Stochastic geometry study on device-to-device communication as a disaster relief solution," *IEEE Trans. Veh. Technol.*, vol. 65, no. 5, pp. 3005–3017, May 2016.
- [94] M. Wang, X. Wang, X. Chai, and W. Huangfu, "The impact of device-todevice communication on the capacity of cellular systems," in *Proc. Int. Conf. Comput., Inf. Telecommun. Syst. (CITS)*, Jul. 2016, pp. 1–5.
- [95] P. Jänis, C.-H. Yu, K. Doppler, C. Ribeiro, and C. Wijting, "Device-todevice communication underlaying cellular communications systems," *Int. J. Commun., Netw. Syst. Sci.*, vol. 2, no. 3, pp. 169–178, Jun. 2009.
- [96] K. Doppler, M. Rinne, C. Wijting, C. B. Ribeiro, and K. Hugl, "Deviceto-device communication as an underlay to LTE-advanced networks," *IEEE Commun. Mag.*, vol. 47, no. 12, pp. 42–49, Dec. 2009.
- [97] Y. Zhang, Y. Yang, and L. Dai, "Energy efficiency maximization for device-to-device communication underlaying cellular networks on multiple bands," *IEEE Access*, vol. 4, pp. 7682–7691, 2016.
- [98] W. Lu and M. Di Renzo, "Stochastic geometry modeling and systemlevel analysis & optimization of relay-aided downlink cellular networks," *IEEE Trans. Commun.*, vol. 63, no. 11, pp. 4063–4085, Nov. 2015.
- [99] K. Ntontin, M. Di Renzo, and C. Verikoukis, "On the feasibility of full-duplex relaying in multiple-antenna cellular networks," *IEEE Trans. Commun.*, vol. 65, no. 5, pp. 2234–2249, Feb. 2017.
- [100] A. P. Buunk and F. X. Gibbons, "Social comparison: The end of a theory and the emergence of a field," Org. Behav. Hum. Decis. Processes, vol. 102, no. 1, pp. 3–21, 2007.
- [101] M. Afshang, H. S. Dhillon, and P. H. J. Chong, "Coverage and area spectral efficiency of clustered device-to-device networks," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2015, pp. 1–6.
- [102] W. Yi, Y. Liu, and A. Nallanathan, "Modeling and analysis of D2D millimeter-wave networks with poisson cluster processes," *IEEE Trans. Commun.*, vol. 65, no. 12, pp. 5574–5588, Dec. 2017.
- [103] M. Afshang and H. S. Dhillon, "Spatial modeling of device-to-device networks: Poisson cluster process meets Poisson hole process," in *Proc.* 49th Asilomar Conf. Signals, Syst. Comput., Nov. 2015, pp. 317–321.

- [104] M. Afshang and H. S. Dhillon, "Optimal geographic caching in finite wireless networks," in *Proc. IEEE 17th Int. Workshop Signal Process. Adv. Wireless Commun. (SPAWC)*, Jul. 2016, pp. 1–5.
- [105] M. Afshang, H. S. Dhillon, and P. H. J. Chong, "Modeling and performance analysis of clustered device-to-device networks," *IEEE Trans. Wireless Commun.*, vol. 15, no. 7, pp. 4957–4972, Jul. 2016.
- [106] M. Afshang, H. S. Dhillon, and P. H. J. Chong, "Fundamentals of clustercentric content placement in cache-enabled device-to-device networks," *IEEE Trans. Commun.*, vol. 64, no. 6, pp. 2511–2526, Jun. 2016.
- [107] S. A. R. Zaidi, M. Ghogho, and D. C. McLernon, "Transmission capacity analysis of cognitive radio networks under co-existence constraints," in *Proc. IEEE 11th Int. Workshop Signal Process. Adv. Wireless Commun.* (SPAWC), Jun. 2010, pp. 1–5.
- [108] T. Jing, X. Chen, Y. Huo, and X. Cheng, "Achievable transmission capacity of cognitive mesh networks with different media access control," in *Proc. IEEE INFOCOM*, Mar. 2012, pp. 1764–1772.
- [109] X. Kang, Y. C. Liang, A. Nallanathan, H. K. Garg, and R. Zhang, "Optimal power allocation for fading channels in cognitive radio networks: Ergodic capacity and outage capacity," *IEEE Trans. Wireless Commun.*, vol. 8, no. 2, pp. 940–950, Feb. 2009.
- [110] C. Yin, L. Gao, T. Liu, and S. Cui, "Transmission capacities for overlaid wireless ad hoc networks with outage constraints," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2009, pp. 1–5.
- [111] A. Hasan and J. G. Andrews, "The guard zone in wireless ad hoc networks," *IEEE Trans. Wireless Commun.*, vol. 6, no. 3, pp. 897–906, Mar. 2007.
- [112] P.-V. Mekikis, A. Antonopoulos, E. Kartsakli, A. S. Lalos, L. Alonso, and C. Verikoukis, "Information exchange in randomly deployed dense WSNs with wireless energy harvesting capabilities," *IEEE Trans. Wireless Commun.*, vol. 15, no. 4, pp. 3008–3018, Apr. 2016.
- [113] H. Jiang, H. Yang, Y. Luo, Q. Zhang, and M. Zeng, "Transmission capacity analysis for underlay relay-assisted energy harvesting cognitive sensor networks," *IEEE Access*, vol. 7, pp. 63778–63788, 2019.
- [114] R. Atat, L. Liu, N. Mastronarde, and Y. Yi, "Energy harvesting-based D2D-assisted machine-type communications," *IEEE Trans. Commun.*, vol. 65, no. 3, pp. 1289–1302, Mar. 2017.
- [115] R. K. Ganti and M. Haenggi, "Spatial and temporal correlation of the interference in ALOHA ad hoc networks," *IEEE Commun. Lett.*, vol. 13, no. 9, pp. 631–633, Sep. 2009.
- [116] T. Kimura and H. Saito, "Temporal correlation of interference under spatially correlated shadowing," in *Proc. 16th Int. Symp. Modeling Optim. Mobile, Ad Hoc, Wireless Netw. (WiOpt)*, May 2018, pp. 1–6.
- [117] S. Krishnan and H. S. Dhillon, "Spatio-temporal interference correlation and joint coverage in cellular networks," *IEEE Trans. Wireless Commun.*, vol. 16, no. 9, pp. 5659–5672, Sep. 2017.
- [118] R. Arshad, L. H. Afify, H. E. Sawy, T. Y. Al-Naffouri, and M. Alouini, "On the effect of uplink power control on temporal retransmission diversity," *IEEE Wireless Commun. Lett.*, vol. 8, no. 1, pp. 309–312, Sep. 2019.
- [119] B. Alawieh, Y. Zhang, C. Assi, and H. Mouftah, "Improving spatial reuse in multihop wireless networks—A survey," *IEEE Commun. Surveys Tuts.*, vol. 11, no. 3, pp. 71–91, 3rd Quart., 2009.
- [120] R. Menon, R. M. Buehrer, and J. H. Reed, "On the impact of dynamic spectrum sharing techniques on legacy radio systems," *IEEE Trans. Wireless Commun.*, vol. 7, no. 11, pp. 4198–4207, Nov. 2008.
- [121] A. Ghasemi and E. S. Sousa, "Fundamental limits of spectrum-sharing in fading environments," *IEEE Trans. Wireless Commun.*, vol. 6, no. 2, pp. 649–658, Feb. 2007.
- [122] Y. Jiang, Q. Liu, F. Zheng, X. Gao, and X. You, "Energy-efficient joint resource allocation and power control for D2D communications," *IEEE Trans. Veh. Technol.*, vol. 65, no. 8, pp. 6119–6127, Aug. 2016.
- [123] F. Librino and M. Zorzi, "Performance of advanced decoding schemes for uplink relaying in cellular networks," *IEEE Trans. Commun.*, vol. 63, no. 1, pp. 79–93, Jan. 2015.
- [124] H. Nourizadeh, S. Nourizadeh, and R. Tafazolli, "Performance evaluation of cellular networks with mobile and fixed relay station," in *Proc. IEEE* 64th Veh. Technol. Conf. (VTC-Fall), Sep. 2006, pp. 1–5.
- [125] L. Wang, T. Peng, Y. Yang, and W. Wang, "Interference constrained relay selection of D2D communication for relay purpose underlaying cellular networks," in *Proc. 8th Int. Conf. WiCOM, Netw. Mobile Comput.*, Sep. 2012, pp. 1–5.
- [126] X. Ma, R. Yin, G. Yu, and Z. Zhang, "A distributed relay selection method for relay assisted device-to-device communication system," in *Proc. IEEE 23rd Int. Symp. Pers. Indoor Mobile Radio Commun. (PIMRC)*, Sep. 2012, pp. 1020–1024.

- [127] R. Madan, N. B. Mehta, A. F. Molisch, and J. Zhang, "Energy-efficient cooperative relaying over fading channels with simple relay selection," *IEEE Trans. Wireless Commun.*, vol. 7, no. 8, pp. 3013–3025, Aug. 2008.
- [128] K. Vanganuru, M. Puzio, G. Sternberg, K. Shah, and S. Kaur, "Uplink system capacity of a cellular network with cooperative mobile relay," in *Proc. Wireless Telecommun. Symp. (WTS)*, Apr. 2011, pp. 1–7.
- [129] T. Q. Duong, V. N. Q. Bao, and H.-J. Zepernick, "Exact outage probability of cognitive AF relaying with underlay spectrum sharing," *Electron. Lett.*, vol. 47, no. 17, pp. 1001–1002, Aug. 2011.
- [130] A. K. Sadek, Z. Han, and K. J. R. Liu, "Distributed relay-assignment protocols for coverage expansion in cooperative wireless networks," *IEEE Trans. Mobile Comput.*, vol. 9, no. 4, pp. 505–515, Apr. 2010.
- [131] Y. Chen, P. Martins, L. Decreusefond, F. Yan, and X. Lagrange, "Stochastic analysis of a cellular network with mobile relays," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2014, pp. 4758–4763.
- [132] P. Lin, Q. Song, Y. Yu, and A. Jamalipour, "Extensive cooperative caching in D2D integrated cellular networks," *IEEE Commun. Lett.*, vol. 21, no. 9, pp. 2101–2104, Sep. 2017.
- [133] M. Ji, G. Caire, and A. F. Molisch, "Fundamental limits of caching in wireless D2D networks," *IEEE Trans. Inf. Theory*, vol. 62, no. 2, pp. 849–869, Feb. 2016.
- [134] K. Zhu, W. Zhi, L. Zhang, X. Chen, and X. Fu, "Social-aware incentivized caching for D2D communications," *IEEE Access*, vol. 4, pp. 7585–7593, 2016.
- [135] Q. Li, Y. Zhang, A. Pandharipande, X. Ge, and J. Zhang, "D2Dassisted caching on truncated zipf distribution," *IEEE Access*, vol. 7, pp. 13411–13421, 2019.
- [136] J. Elias and B. Błaszczyszyn, "Optimal geographic caching in cellular networks with linear content coding," in Proc. 15th Int. Symp. Modeling Optim. Mobile, Ad Hoc, Wireless Netw. (WiOpt), May 2017, pp. 1–6.
- [137] Y. Keshtkarjahromi, H. Seferoglu, R. Ansari, and A. Khokhar, "Deviceto-device networking meets cellular via network coding," *IEEE/ACM Trans. Netw.*, vol. 26, no. 1, pp. 370–383, Feb. 2018.
- [138] P. V. Mekikis, A. S. Lalos, A. Antonopoulos, L. Alonso, and C. Verikoukis, "Wireless energy harvesting in two-way network coded cooperative communications: A stochastic approach for large scale networks," *IEEE Commun. Lett.*, vol. 18, no. 6, pp. 1011–1014, Jun. 2014.
- [139] P. Wang, G. Mao, Z. Li, X. Ge, and B. D. O. Anderson, "Network coding based wireless broadcast with performance guarantee," *IEEE Trans. Wireless Commun.*, vol. 14, no. 1, pp. 532–544, Jan. 2015.
- [140] R. Vaze and R. W. Heath, Jr., "On the capacity and diversity-multiplexing tradeoff of the two-way relay channel," *IEEE Trans. Inf. Theory*, vol. 57, no. 7, pp. 4219–4234, Jul. 2011.
- [141] G. Liu, F. R. Yu, H. Ji, V. C. M. Leung, and X. Li, "In-band full-duplex relaying: A survey, research issues and challenges," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 2, pp. 500–524, 2nd Quart., 2015.
- [142] C. Sun and C. Yang, "Is two-way relay more energy efficient?" in Proc. IEEE Global Telecommun. Conf. (GLOBECOM), Dec. 2011, pp. 1–6.
- [143] H. Yu, Y. Li, X. Zhong, L. Wang, and J. Wang, "The analysis of the energy efficiency for the decode-and-forward two-way relay networks," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2013, pp. 2823–2827.
- [144] R. Vaze, K. T. Truong, S. Weber, and R. W. Heath, Jr., "Two-way transmission capacity of wireless ad-hoc networks," *IEEE Trans. Wireless Commun.*, vol. 10, no. 6, pp. 1966–1975, Jun. 2011.
- [145] H. Cao, L. Fu, and H. Dai, "Throughput analysis of the two-way relay system with network coding and energy harvesting," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2017, pp. 1–6.
- [146] W. Li, M.-L. Ku, Y. Chen, K. J. R. Liu, and S. Zhu, "Performance analysis for two-way network-coded dual-relay networks with stochastic energy harvesting," *IEEE Trans. Wireless Commun.*, vol. 16, no. 9, pp. 5747–5761, Sep. 2017.
- [147] D. Wang, R. Zhang, X. Cheng, L. Yang, and C. Chen, "Relay selection in full-duplex energy-harvesting two-way relay networks," *IEEE Trans. Green Commun. Netw.*, vol. 1, no. 2, pp. 182–191, Jun. 2017.
- [148] A. Alsharoa, H. Ghazzai, A. E. Kamal, and A. Kadri, "Optimization of a power splitting protocol for two-way multiple energy harvesting relay system," *IEEE Trans. Green Commun. Netw.*, vol. 1, no. 4, pp. 444–457, Dec. 2017.
- [149] R. Vahidnia, A. Anpalagan, and J. Mirzaei, "Relay selection in energy harvesting two-way communication networks," in *Proc. IEEE 26th Annu. Int. Symp. Pers., Indoor, Mobile Radio Commun. (PIMRC)*, Aug. 2015, pp. 966–970.

- [150] D. Jiang, H. Zheng, D. Tang, and Y. Tang, "Relay selection and power allocation for cognitive energy harvesting two-way relaying networks," in *Proc. IEEE 5th Int. Conf. Electron. Inf. Emergency Commun. (ICEIEC)*, May 2015, pp. 163–166.
- [151] W. Li, M.-L. Ku, Y. Chen, and K. J. R. Liu, "On outage probability for two-way relay networks with stochastic energy harvesting," *IEEE Trans. Commun.*, vol. 64, no. 5, pp. 1901–1915, May 2016.
- [152] P. Nguyen-Huu and K. Ho-Van, "Bidirectional relaying with energy harvesting capable relay: Outage analysis for Nakagami-*m* fading," *Telecommun. Syst.*, vol. 69, no. 3, pp. 335–347, 2018.
- [153] S. Javadi and E. Soleimani-Nasab, "Two-way interference-limited af relaying with wireless power transfer," in *Proc. 24th Telecommun. Forum* (*TELFOR*), Nov. 2016, pp. 1–4.
- [154] X. Zhu, N.-P. Nguyen, T. T. Lam, and D.-B. Ha, "Throughput analysis of bidirectional relaying networks with wireless power transfer over Nakagami fading," in *Proc. Int. Conf. Commun., Manage. Telecommun.* (*ComManTel*), Dec. 2015, pp. 153–156.
- [155] J. J. Park, J. H. Moon, and D. I. Kim, "Time-switching based in-band full duplex wireless powered two-way relay," in *Proc. URSI Asia–Pacific Radio Sci. Conf. (URSI AP-RASC)*, Aug. 2016, pp. 438–441.
- [156] A. Salem and K. A. Hamdi, "Wireless power transfer in multi-pair twoway AF relaying networks," *IEEE Trans. Commun.*, vol. 64, no. 11, pp. 4578–4591, Nov. 2016.
- [157] Z. Chen, B. Xia, and H. Liu, "Wireless information and power transfer in two-way amplify-and-forward relaying channels," in *Proc. IEEE Global Conf. Signal Inf. Process. (GlobalSIP)*, Dec. 2014, pp. 168–172.
- [158] J. G. Andrews, "Interference cancellation for cellular systems: A contemporary overview," *IEEE Wireless Commun.*, vol. 12, no. 2, pp. 19–29, Apr. 2005.
- [159] Y. Wang, Q. Cui, M. Haenggi, and Z. Tan, "On the SIR meta distribution for Poisson networks with interference cancellation," *IEEE Wireless Commun. Lett.*, vol. 7, no. 1, pp. 26–29, Feb. 2018.
- [160] P. Cardieri and J. M. C. Brito, "Packet retransmission in D2D underlay cellular networks," *IEEE Commun. Lett.*, vol. 22, no. 9, pp. 1914–1917, Jul. 2018.
- [161] Y. J. Chun, S. L. Cotton, H. S. Dhillon, F. J. Lopez-Martinez, J. F. Paris, and K. S. Yoo, "A comprehensive analysis of 5G heterogeneous cellular systems operating over κ-μ shadowed fading channels," *IEEE Trans. Wireless Commun.*, vol. 16, no. 11, pp. 6995–7010, Nov. 2017.
- [162] S. Huang, Z. Wei, X. Yuan, Z. Feng, and P. Zhang, "Performance characterization of machine-to-machine networks with energy harvesting and social-aware relays," *IEEE Access*, vol. 5, pp. 13297–13307, 2017.
- [163] Q. Yang, K. Lu, V. Mancuso, and C.-H. Youn, "Device-to-device communications with social awareness [guest editorial]," *IEEE Wireless Commun.*, vol. 23, no. 4, pp. 10–11, Aug. 2016.
- [164] Y. Li, T. Wu, P. Hui, D. Jin, and S. Chen, "Social-aware D2D communications: Qualitative insights and quantitative analysis," *IEEE Commun. Mag.*, vol. 52, no. 6, pp. 150–158, Jun. 2014.
- [165] J. Yan, D. Wu, C. Zhang, H. Wang, and R. Wang, "Socially aware D2D cooperative communications for enhancing Internet of Things application," *EURASIP J. Wireless Commun. Netw.*, vol. 2018, no. 1, p. 132, 2018.
- [166] X. Chen, B. Proulx, X. Gong, and J. Zhang, "Exploiting social ties for cooperative D2D communications: A mobile social networking case," *IEEE/ACM Trans. Netw.*, vol. 23, no. 5, pp. 1471–1484, Oct. 2015.
- [167] Y. Gong, G. Chen, and T. Xie, "Using buffers in trust-aware relay selection networks with spatially random relays," *IEEE Trans. Wireless Commun.*, vol. 17, no. 9, pp. 5818–5826, Jul. 2018.
- [168] H. Mao, W. Feng, Y. Zhao, and N. Ge, "Joint social-position relationship based cooperation among mobile terminals," *IEEE Commun. Lett.*, vol. 18, no. 12, pp. 2165–2168, Dec. 2014.
- [169] M. Mozaffari, W. Saad, M. Bennis, and M. Debbah, "Wireless communication using unmanned aerial vehicles (UAVs): Optimal transport theory for hover time optimization," *IEEE Trans. Wireless Commun.*, vol. 16, no. 12, pp. 8052–8066, Dec. 2017.
- [170] S. A. R. Naqvi, S. A. Hassan, H. Pervaiz, and Q. Ni, "Drone-aided communication as a key enabler for 5G and resilient public safety networks," *IEEE Commun. Mag.*, vol. 56, no. 1, pp. 36–42, Jan. 2018.
- [171] Y. Zeng, R. Zhang, and T. J. Lim, "Wireless communications with unmanned aerial vehicles: Opportunities and challenges," *IEEE Commun. Mag.*, vol. 54, no. 5, pp. 36–42, May 2016.
- [172] H. Wang, J. Chen, G. Ding, and S. Wang, "D2D communications underlaying UAV-assisted access networks," *IEEE Access*, vol. 6, pp. 46244–46255, 2018.

- [173] V. V. Chetlur and H. S. Dhillon, "Downlink coverage analysis for a finite 3-D wireless network of unmanned aerial vehicles," *IEEE Trans. Commun.*, vol. 65, no. 10, pp. 4543–4558, Jul. 2017.
- [174] F. Lagum, I. Bor-Yaliniz, and H. Yanikomeroglu, "Strategic densification with UAV-BSs in cellular networks," *IEEE Wireless Commun. Lett.*, vol. 7, no. 3, pp. 384–387, Jun. 2018.
- [175] R. Arshad, L. Lampe, H. ElSawy, and M. Hossain, "Integrating UAVs into existing wireless networks: A stochastic geometry approach," 2018, arXiv:1810.07801. [Online]. Available: https://arxiv.org/abs/1810.07801
- [176] I. Bor-Yaliniz and H. Yanikomeroglu, "The new frontier in RAN heterogeneity: Multi-tier drone-cells," *IEEE Commun. Mag.*, vol. 54, no. 11, pp. 48–55, Nov. 2016.
- [177] W. Yi, Y. Liu, A. Nallanathan, and G. K. Karagiannidis, "A unified spatial framework for clustered UAV networks based on stochastic geometry," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2018, pp. 1–6.
- [178] I. Krikidis, "Relay selection in wireless powered cooperative networks with energy storage," *IEEE J. Sel. Area Commun.*, vol. 33, no. 12, pp. 2596–2610, Dec. 2015.
- [179] H. H. Yang, J. Lee, and T. Q. S. Quek, "Heterogeneous cellular network with energy harvesting-based D2D communication," *IEEE Trans. Wireless Commun.*, vol. 15, no. 2, pp. 1406–1419, Feb. 2016.
- [180] C. Zhai, J. Liu, and L. Zheng, "Cooperative spectrum sharing with wireless energy harvesting in cognitive radio networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 7, pp. 5303–5316, Jul. 2016.
- [181] Z. Yan, S. Chen, X. Zhang, and H. L. Liu, "Outage performance analysis of wireless energy harvesting relay-assisted random underlay cognitive networks," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 2691–2699, Aug. 2018.
- [182] B. L. Pham and A.-V. Pham, "Triple bands antenna and high efficiency rectifier design for RF energy harvesting at 900, 1900 and 2400 MHz," in *IEEE MTT-S Int. Microw. Symp. Dig.*, Jun. 2013, pp. 1–3.
- [183] S. Kusaladharma and C. Tellambura, "Interference and outage in random D2D networks under millimeter wave channels," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2017, pp. 1–7.
- [184] S. Wu, R. Atat, N. Mastronarde, and L. Liu, "Coverage analysis of D2D relay-assisted millimeter-wave cellular networks," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Mar. 2017, pp. 1–6.
- [185] S. Wu and N. Mastronarde, "Coverage and spectral efficiency of deviceto-device relay-assisted cellular networks," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2018, pp. 1–6.
- [186] K. Belbase, Z. Zhang, H. Jiang, and C. Tellambura, "Coverage analysis of millimeter wave decode-and-forward networks with best relay selection," *IEEE Access*, vol. 6, pp. 22670–22683, 2018.
- [187] N. Giatsoglou, K. Ntontin, E. Kartsakli, A. Antonopoulos, and C. Verikoukis, "D2D-aware device caching in mmWave-cellular networks," *IEEE J. Sel. Areas Commun.*, vol. 35, no. 9, pp. 2025–2037, Sep. 2017.
- [188] C. Ma, J. Liu, X. Tian, H. Yu, Y. Cui, and X. Wang, "Interference exploitation in D2D-enabled cellular networks: A secrecy perspective," *IEEE Trans. Commun.*, vol. 63, no. 1, pp. 229–242, Jan. 2015.
- [189] Y. Liu, L. Wang, S. A. R. Zaidi, M. Elkashlan, and T. Q. Duong, "Secure D2D communication in large-scale cognitive cellular networks: A wireless power transfer model," *IEEE Trans. Commun.*, vol. 64, no. 1, pp. 329–342, Jan. 2016.
- [190] Y. J. Tolossa, S. Vuppala, G. Kaddoum, and G. Abreu, "On the uplink secrecy capacity analysis in D2D-enabled cellular network," *IEEE Syst. J.*, vol. 12, no. 3, pp. 2297–2307, 2018.
- [191] Y. J. Chun, S. L. Cotton, H. S. Dhillon, A. Ghrayeb, and M. O. Hasna, "A stochastic geometric analysis of device-to-device communications operating over generalized fading channels," *IEEE Trans. Wireless Commun.*, vol. 16, no. 7, pp. 4151–4165, Jul. 2017.
- [192] I. Trigui and S. Affes, "Unified analysis and optimization of D2D communications in cellular networks over fading channels," *IEEE Trans. Commun.*, vol. 67, no. 1, pp. 724–736, Jul. 2018.
- [193] M. Gharbieh, H. ElSawy, A. Bader, and M.-S. Alouini, "Spatiotemporal stochastic modeling of IoT enabled cellular networks: Scalability and stability analysis," *IEEE Trans. Commun.*, vol. 65, no. 8, pp. 3585–3600, Aug. 2017.
- [194] H. Chen, L. Liu, N. Mastronarde, L. Ma, and Y. Yi, "Cooperative retransmission for massive MTC under spatiotemporally correlated interference," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2016, pp. 1–6.

- [195] Z. Ding, X. Lei, G. K. Karagiannidis, R. Schober, J. Yuan, and V. Bhargava, "A survey on non-orthogonal multiple access for 5G networks: Research challenges and future trends," *IEEE J. Sel. Areas Commun.*, vol. 35, no. 10, pp. 2181–2195, Oct. 2017.
- [196] Z. Shi, S. Ma, H. E. Sawy, G. Yang, and M.-S. Alouini, "Cooperative HARQ-assisted NOMA scheme in large-scale D2D networks," *IEEE Trans. Commun.*, vol. 66, no. 9, pp. 4286–4302, Sep. 2018.
- [197] K. S. Ali, M. Haenggi, H. El Sawy, A. Chaaban, and M.-S. Alouini, "Downlink non-orthogonal multiple access (NOMA) in Poisson networks," *IEEE Trans. Commun.*, vol. 67, no. 2, pp. 1613–1628, Feb. 2019.
- [198] K. S. Ali, H. El Sawy, and M.-S. Alouini, "Meta distribution of downlink non-orthogonal multiple access (NOMA) in Poisson networks," *IEEE Wireless Commun. Lett.*, vol. 8, no. 2, pp. 572–575, Apr. 2019.
- [199] T. Hou, Y. Liu, Z. Song, X. Sun, and Y. Chen, "Multiple antenna aided NOMA in UAV networks: A stochastic geometry approach," *IEEE Trans. Commun.*, vol. 67, no. 2, pp. 1031–1044, Feb. 2019.
- [200] Y. Liu, Z. Qin, Y. Cai, Y. Gao, G. Y. Li, and A. Nallanathan, "UAV communications based on non-orthogonal multiple access," *IEEE Wireless Commun.*, vol. 26, no. 1, pp. 52–57, Feb. 2019.
- [201] Z. Zhao, M. Xu, W. Xie, Z. Ding, and G. K. Karagiannidis, "Coverage performance of noma in wireless caching networks," *IEEE Commun. Lett.*, vol. 22, no. 7, pp. 1458–1461, Apr. 2018.
- [202] Y. Liu, Z. Ding, M. Elkashlan, and J. Yuan, "Nonorthogonal multiple access in large-scale underlay cognitive radio networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 12, pp. 10152–10157, Feb. 2016.
- [203] X. Yue, Y. Liu, S. Kang, A. Nallanathan, and Z. Ding, "Spatially random relay selection for full/half-duplex cooperative NOMA networks," *IEEE Trans. Commun.*, vol. 66, no. 8, pp. 3294–3308, Aug. 2018.
- [204] Y. Sun, Z. Ding, and X. Dai, "On the performance of downlink NOMA in multi-cell mmwave networks," *IEEE Commun. Lett.*, vol. 22, no. 11, pp. 2366–2369, Sep. 2018.
- [205] A. Anwar, B.-C. Seet, S. F. Hasan, X. J. Li, P. H. J. Chong, and M. Y. Chung, "An analytical framework for multi-tier NOMA networks with underlay D2D communications," *IEEE Access*, vol. 6, pp. 59221–59241, 2018.
- [206] A. Anwar, B.-C. Seet, and X. Li, "Quality of service based NOMA group D2D communications," *Future Internet*, vol. 9, no. 4, p. 73, 2017.
- [207] X. Yu, Q. Cui, and M. Haenggi, "SIR meta distribution for spatiotemporal cooperation in Poisson cellular networks," *IEEE Access*, to be published.
- [208] M. Salehi, H. Tabassum, and E. Hossain, "Meta distribution of SIR in large-scale uplink and downlink NOMA networks," *IEEE Trans. Commun.*, vol. 67, no. 4, pp. 3009–3025, Apr. 2019.
- [209] M. Salehi, A. Mohammadi, and M. Haenggi, "Analysis of D2D underlaid cellular networks: SIR meta distribution and mean local delay," *IEEE Trans. Commun.*, vol. 65, no. 7, pp. 2904–2916, Jul. 2017.
- [210] M. Haenggi, "The meta distribution of the SIR in Poisson bipolar and cellular networks," *IEEE Trans. Wireless Commun.*, vol. 15, no. 4, pp. 2577–2589, Apr. 2016.
- [211] Y. Wang, M. Haenggi, and Z. Tan, "The meta distribution of the sir for cellular networks with power control," *IEEE Trans. Commun.*, vol. 66, no. 4, pp. 1745–1757, Apr. 2018.
- [212] Q. Cui, X. Yu, Y. Wang, and M. Haenggi, "The SIR meta distribution in Poisson cellular networks with base station cooperation," *IEEE Trans. Commun.*, vol. 66, no. 3, pp. 1234–1249, Mar. 2018.
- [213] N. Deng and M. Haenggi, "The meta distribution of the SINR in mmwave D2D networks," in *Proc. IEEE Global Commun. Conf. (GLOBE-COM)*, Dec. 2017, pp. 1–6.
- [214] S. S. Kalamkar and M. Haenggi, "Spatial outage capacity of Poisson bipolar networks," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2017, pp. 1–6.
- [215] S. Joshi and R. K. Mallik, "Coverage and interference in D2D networks with Poisson cluster process," *IEEE Commun. Lett.*, vol. 22, no. 5, pp. 1098–1101, May 2018.
- [216] S. M. Azimi-Abarghouyi, B. Makki, M. Haenggi, M. Nasiri-Kenari, and T. Svensson, "Stochastic geometry modeling and analysis of single- and multi-cluster wireless networks," *IEEE Trans. Commun.*, vol. 66, no. 10, pp. 4981–4996, Oct. 2018.
  [217] Y. Xu, "On the performance of device-to-device communications
- [217] Y. Xu, "On the performance of device-to-device communications with delay constraint," *IEEE Trans. Veh. Technol.*, vol. 65, no. 11, pp. 9330–9344, Nov. 2016.
- [218] L. Chen, C. Liu, X. Hong, C.-X. Wang, J. Thompson, and J. Shi, "Capacity and delay tradeoff of secondary cellular networks with spectrum aggregation," *IEEE Trans. Wireless Commun.*, vol. 17, no. 6, pp. 3974–3987, Jun. 2018.

- [219] W. Nie, Y. Zhong, F.-C. Zheng, W. Zhang, and T. O'Farrell, "Hetnets with random DTX scheme: Local delay and energy efficiency," *IEEE Trans. Veh. Technol.*, vol. 65, no. 8, pp. 6601–6613, Sep. 2016.
- [220] H. Ding, X. Wang, D. B. da Costa, and J. Ge, "Interference modeling in clustered device-to-device networks with uniform transmitter selection," *IEEE Trans. Wireless Commun.*, vol. 16, no. 12, pp. 7906–7918, Dec. 2017.
- [221] M. Amjad, F. Akhtar, M. H. Rehmani, M. Reisslein, and T. Umer, "Fullduplex communication in cognitive radio networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 4, pp. 2158–2191, 4th Quart., 2017.
- [222] D. Kim, H. Lee, and D. Hong, "A survey of in-band full-duplex transmission: From the perspective of PHY and MAC layers," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2017–2046, 4th quart., 2015.
- [223] S. K. Sharma, T. E. Bogale, L. B. Le, S. Chatzinotas, X. Wang, and B. Ottersten, "Dynamic spectrum sharing in 5G wireless networks with full-duplex technology: Recent advances and research challenges," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 1, pp. 674–707, 1st Quart., 2018.
- [224] X. Chai, T. Liu, C. Xing, H. Xiao, and Z. Zhang, "Throughput improvement in cellular networks via full-duplex based device-to-device communications," *IEEE Access*, vol. 4, pp. 7645–7657, 2016.
- [225] L. Wang, F. Tian, T. Svensson, D. Feng, M. Song, and S. Li, "Exploiting full duplex for device-to-device communications in heterogeneous networks," *IEEE Commun. Mag.*, vol. 53, no. 5, pp. 146–152, May 2015.
- [226] N. Haider, E. Dutkiewicz, D. N. Nguyen, M. Mueck, and S. Srikanteswarae, "The impact on full duplex D2D communication of different LTE transmission techniques," in *Proc. IEEE 85th Veh. Technol. Conf. (VTC Spring)*, Jun. 2017, pp. 1–5.
- [227] K. S. Ali, H. ElSawy, and M.-S. Alouini, "Modeling cellular networks with full-duplex D2D communication: A stochastic geometry approach," *IEEE Trans. Commun.*, vol. 64, no. 10, pp. 4409–4424, Oct. 2016.
- [228] S. Badri, M. Naslcheraghi, and M. Rasti, "Performance analysis of joint pairing and mode selection in D2D communications with FD radios," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2018, pp. 1–6.
- [229] T.-C. Hou and V. O. K. Li, "Transmission range control in multihop packet radio networks," *IEEE Trans. Commun.*, vol. 34, no. 1, pp. 38–44, Jan. 1986.
- [230] P. H. J. Nardelli, P. Cardieri, and M. Latva-Aho, "Efficiency of wireless networks under different hopping strategies," *IEEE Trans. Wireless Commun.*, vol. 11, no. 1, pp. 15–20, Jan. 2012.
- [231] M. J. Farooq, H. ElSawy, Q. Zhu, and M.-S. Alouini, "Optimizing mission critical data dissemination in massive IoT networks," in *Proc.* 15th Int. Symp. Modeling Optim. Mobile, Ad Hoc, Wireless Netw. (WiOpt), May 2017, pp. 1–6.
- [232] V. Wong and C. Leung, "Effect of Rayleigh fading in a multihop mobile packet radio network with capture," *IEEE Trans. Veh. Technol.*, vol. 44, no. 3, pp. 630–637, Aug. 1995.
- [233] M. J. Farooq, H. ElSawy, and M.-S. Alouini, "A stochastic geometry model for multi-hop highway vehicular communication," *IEEE Trans. Wireless Commun.*, vol. 15, no. 3, pp. 2276–2291, Mar. 2016.



**OLUWATOSIN AHMED AMODU** received the B.Tech. degree in electrical electronics engineering from The Federal University of Technology at Akure, Akure, Nigeria, in 2012, and the master's degree in computer science with a specialization in distributed computing with Universiti Putra Malaysia, in 2016, where he is currently pursuing the Ph.D. degree in wireless communication and network engineering. His research interests include sensor networks, machine-type communi-

cations, device-to-device communication, spectrum sharing networks, and stochastic geometry.



**MOHAMED OTHMAN** received the Ph.D. degree (Hons.) from Universiti Kebangsaan Malaysia. He is currently a Professor in computer science with the Department of Communication Technology and Network, Universiti Putra Malaysia (UPM). Prior to that, he was the Deputy Director of the Information Development and Communication Center, where he was the in charge of UPM Net network campus, the uSport Wireless Communication Project, and the UPM DataCenter. He is also

an Associate Researcher and a Coordinator of high-speed machines with the Laboratory of Computational Science and Informatics, Institute of Mathematical Science, UPM. In 2017, he received an Honorary Professor with Silkway International University (SWIU), Shymkent, Kazakhstan. He was also a Visiting Professor with South Kazakhstan State University, Shymkent, and L. N. Gumilyov Eurasian National University, Astana, Kazakhstan. He has published more than 300 international journals and 330 proceeding papers. He has also filed six Malaysian, one Japanese, one South Korean, and three U.S. patents. His main research interests include the fields of computer networks, parallel and distributed computing, highspeed interconnection networks, network design and management (network security and wireless and traffic monitoring), consensus in the IoT, and mathematical modelling in scientific computing. He is a Life Member of the Malaysian National Computer Confederation and the Malaysian Mathematical Society. He was a recipient of the Best Ph.D. Thesis from Sime Darby Malaysia and the Malaysian Mathematical Science Society, in 2000.

**NOR KAMARIAH NOORDIN** received the B.Sc. degree in electrical engineering with a major in telecommunications from The University of Alabama, Tuscaloosa, AL, USA, in 1987, the master's degree from University Teknologi Malaysia, and the Ph.D. degree from University Putra Malaysia (UPM). She has been with UPM, since 1988. She was appointed as an Associate Professor, in 2006, and a Professor, in 2012. During her tenure, she has been the head of department, the deputy dean of academics, the director of corporate strategy and communication of the university. She is currently the Dean of the Faculty of Engineering, UPM. During more than 25 years with the university, she has published more than 200 journals and conference papers and has secured more than 30 research and consultancy projects.

**IDAWATY AHMAD** received the bachelor's and master's degrees in information science from Saga University, Japan, and the Ph.D. degree in computer networks from Universiti Putra Malaysia, where she has been a Lecturer with the Faculty of Computer Science and Information Technology, since 2000. She specializes in the areas of real-time systems, network protocols, and simulation/modeling, which are core areas in computer science.

. . .